CEO Social Capital and Discretionary Accruals

William R. McCumber•

Huan (Ken) Qiu[•]

Abstract

This study investigates the relationship between CEO social capital and the use of discretionary accruals in a firm, as well as the pricing of discretionary accruals from the capital market, for the period of 1998-2017. We find that firms managed by CEOs with higher social capital are more likely to use discretionary accruals (predicted by cross-sectional Modified Jones (1991) Model) to manage earnings upward, rather than alter firms' real operating activities. However, those behaviors concentrate in the income decreasing (negative) discretionary accruals sub-sample. The result is robust with the control for self-selections bias, the omitted variables, and the endogeneity. Further analysis unveils that compared with the ones used in the firms managed by CEOs with higher social capital, especially the ones in the income decreasing (negative) discretionary accruals used in the firms managed by CEOs with higher social capital, especially the ones in the income decreasing (negative) discretionary accruals used are thus positively priced by the capital market.

JEL Codes: G30, G40, M41

Keywords: CEOs social capital, earnings management, discretionary accruals, pricing of accruals

[•] Associate Professor and JPJ Investments Endowed Professor of Finance, College of Business, Louisiana Tech University, Ruston, Louisiana 71272, United States. mccumber@latech.edu.

^{*} Assistant Professor of Finance, Else School of Management, Millsaps College, Jackson, Mississippi 39210, United States. qiuh@millsaps.edu. Corresponding Author.

CEO Social Capital and Discretionary Accruals

Section 1. Introduction

Literature has long debate on whether accrual-earnings or cash flow has more incremental information content towards a firm's future economic performance (Bowen, Burghstahler, and Daley, 1987; Dechow 1994; Dechow, Kothari, and Watts, 1998; Subramanyam, 1996). Accrual-earnings is viewed as a superior measurement of a firm's performance to cash flow as it helps mitigate the timing and mismatching problems inherent in measuring cash flow (Dechow, 1994). However, due to the flexibility under the Generally Accepted Accounting Principles (GAAP), the choice and magnitude of accruals used in financial reporting is subject to managerial discretion. On the one hand, managerial discretion can enhance the informativeness of accrual-earnings by allowing better communication of private information to the public and can therefore improve the ability of accruals to reflect the economic value of a firm. (Watts and Zimmerman, 1990; Holthausen, 1990; Healy and Palepu, 1993). On the other hand, due to the potential conflict of interest between managers and stockholders, managers may have incentives to use discretionary accruals to opportunistically manage earnings to benefit themselves, and thereby create perversions in the reported earnings (Watts and Zimmerman, 1986; Healy and Palepu, 1993).¹ Since investors can not directly observe the managers' opportunistic behaviors, in turn, they may require a higher expected return to invest in a firm with higher accruals and that leads to a potential higher cost of equity for the firm (Francis, Lafond, Olsson, and Schipper, 2005). As a result, Guay, Kothari, and Watts (1996) suggest that

¹ That is why the use of discretionary accruals is sometimes referred to as the accruals-based earnings management. In this study, these two concepts are interchangeable.

researchers take managers' incentives into account when studying the role of discretionary accruals.

Literature also documents evidence to support that CEO social network is an important channel for business information flow and sharing (Nohria, 1992; Burt, 1997; El-khatib, Fogel, and Jandik, 2015; Fracassi and Tate, 2012; Woolcook, 1998). As the network connection allows CEOs to have easier access to information and resources within the network, CEO network connection is normally considered as a great resource, or more formally, an intangible asset, to a firm. Woolcock (1998) provides a more formal definition, CEO social capital, to measure the importance and influence of CEO within the network. Extant literature reveals that CEO social capital can encourage both reputation-incentive and rent-seeking behaviors through the information and reputation channels, and therefore represents a double-edged sword in terms of firm and market outcomes (El-Khatib et al., 2015; Bhandari, Mammadov, Shelton, and Thevenot, 2018; Griffin, Hong, Liu, and Ryou, 2021; Ferris, Javakhadze, and Raikovic, 2017). Specifically, a growing literature reveals that CEO social capital can positively or negatively influence a firm's overall information environment and financial reporting practices (Bhandari et al., 2018; Griffin et al., 2021), potentially including a firm's decision to use accruals in the financial reporting. Thus, we argue that CEO social capital can influence CEOs' incentives to manage earnings by accruals,² to either opportunistically manage earnings or provide better private information to the public. We are thereby interested in studying the relation between CEO social capital and discretionary accruals. Specifically, we emphasize on examining (1) whether CEO social capital affects the use of discretionary accruals; 3 (2) what type of

² The examples of CEO's incentives to manage earnings by accruals include maintaining their reputation within the labor market and increase their compensation by meeting or beating earnings target.

³ To be consistent, in this study, we assume that firms use discretionary accruals to manage earnings upward by default.

discretionary accruals (positive/income increasing vs. negative/income decreasing) CEO social capital has a greater impact on if CEO social capital affects the use of discretionary accruals; and (3) how CEO social capital influences the pricing of discretionary accruals (by the capital market).

To further study this topic, we first collect C-Suite employment histories and other biographical data on U.S. CEOs from BoardEx to construct CEO network centrality variables to proxy for CEO social capital for the period of 1998-2017.⁴ Specifically, we construct degree (*Degree*) and eigenvector centrality (*Eigenvector*) to capture the number of connections and the connection to the connected people. Additionally, we use principal component analysis to create a third centrality measurement, *PCA*, to capture the common features between *Degree* and *Eigenvector*. Next, we follow extant literature (Jones, 1991; Dechow, Sloan, and Sweeney, 1995; Subramanyam, 1996; Krishnan, 2003) and construct the variable to proxy for the level of discretionary accruals used in a firm (*Dis Accruals*) using cross-sectional Modified Jones (1991) Model.⁵ Furthermore, we collect other firm-level financial and return information from Compustat and CRSP to construct other controls variables as needed. Finally, we merge all the datasets and exclude firm-year observations in utility industries (with SIC codes 4000-4999) and financial companies (SIC codes 6000-6999) and the ones with missing value for required variables in our empirical analysis.

We first find that firms managed by CEOs with higher social capital are more likely to use discretionary accruals to manage earnings upward, rather than alter real operating activities,

⁴ We argue that our CEO network centrality measurements can capture the importance and influence of a CEO within the network, the essential meaning presented by CEO social capital.

⁵ As extant literature reports the substitute effect between the use of accrual-based and real-activities earnings management (e.g., Zang, 2012). To capture the substitute effect, we control for the level of real earnings management used a firm in the main regression and thereby also create a variable to proxy for real earnings management (*Real EM*) following Roychowdhury (2006). In this study, managing earnings through the alteration of real operating activities is also referred to as the real-activities or real earnings management.

after controlling for factors that partially determine the use of earnings management including the substitute effect between these two methods (Zang, 2012; Chan, Chen, Chen, and Yu, 2015; Black, Joo, and Schmarkebeck, 2017). Furthermore, when we partition the sample into subsamples with income increasing (positive) and income decreasing (negative) discretionary accruals, we discover that firms managed by CEOs with higher social capital are more likely to manage earnings upward in the negative discretionary accruals sub-sample. The result is robust with the control for self-selections bias, the omitted variables, and the endogeneity. Moreover, we analyze whether CEO social capital influences the pricing of discretionary accruals. The analyses reveal that compared with the ones used in the firms managed by CEOs with higher social capital, the discretionary accruals used in the firms managed by CEOs with higher social capital have better predictabilities on the firms' future performance⁶ and are thus positively priced by the capital market (proxied for by the increasing relevance of the discretionary accruals to the future stock return of the firms).

Our findings have several contributions to literature. First, we complement the behavioral literature by documenting evidence that CEO social capital influences a firm's use of discretionary accruals. As a firm's use of discretionary accruals reflects the firm's financial reporting practice, our result adds additional evidence to support that CEO social capital help facilitate a firm's financial reporting practice (Bhandari et al., 2018; Griffin et al., 2021).

It is also worth noting that our study is related to the one conducted by Griffin et al. (2021), but our CEO social capital measurement differs from them by focusing on the current board overlapping and by including the connections to the connected people in addition to

⁶ Following extant literature (e.g., Robin and Wu, 2015), we measure firms' future performance by the one-year ahead earnings and the probability of future dividend increase.

number of connections, to capture the hierarchy of social network.⁷ Additionally, our results (regarding the managerial choice of managing earnings upward through discretionary accruals or through the alternation of real activities) are completely opposite to theirs.

Furthermore, we shed light on the choice of discretionary accruals by firms managed by CEOs with higher social capital. The use of discretionary accruals has normally been considered to induce higher potential cost to firms and can be easily detected by SEC (Zang, 2012; Chan et al., 2015; Black et al. 2017; and Griffin et al., 2021), so firms try to avoid using it or shift to managing earnings by altering real operating activities, especially in post-SOX period. However, our evidence reveals that firms managed by CEOs with higher social capital can use discretionary accruals in a positive fashion to convey information to market participants. Contrary to extant literature (Lee, Li, and Yue, 2006; Robin and Wu, 2015), our evidence also suggests that managers can also report income decreasing (negative) discretionary accruals to provide useful private information to the market. Overall, our study provides evidence that the use of discretionary accruals to manage earnings upward in a firm may not always bring in detrimental consequences for the firm. To some extent, the use of discretionary accruals can be beneficial to a firm by improving its informational environment.

Finally, consistent with the extant literature (Diamond, 1985; Faleye, Kovacs, and Venkateswaran, 2014; Ferris et al., 2017), we confirm that firms managed by CEOs with higher social capital have flexibility to take risk (proxied for by choosing to use discretionary accruals to manage earnings upward), however, the risk-taking won't cause bad consequences. To some extent, such risk-taking can benefit the firm by improving its overall informational environment through the better communication of private information to the public.

⁷ El-Khatib et al. (2015) argue that the hierarchy of social network affects the direction of information flow and can therefore affect firm and market outcomes.

The rest of the paper will be organized in the following format: Section 2 discusses previous literatures and develop hypotheses; section 3 introduces key variables and sample construction; section 4 introduces the research design and presents the main empirical results; section 5 concludes.

Section 2. Literature Review and Hypotheses Development

2.1. CEO Social Capital and the use of discretionary accruals

Whether firms operate efficiently is a debatable topic between traditional economic theories (e.g., Neoclassical economic theory; agency theory proposed by Fama, 1980) and modern behavior theories (e.g., upper echelons theory from Hambrick and Mason, 1984). Unlike the traditional economic theories, behavior theories argue against firm's operational efficiency, specifically, the theories assert that executives are heterogeneous in nature and that the characteristics of the executives can potentially influence firm and market outcomes.

Woolcock (1998) first introduces the concept of CEO social capital as a measurement of the power and influence of a CEO within the social network. According to social network theories, CEO social capital is a valuable resource for a firm as it allows its CEO to have easier access to information and resource within the network. In addition, network can enable trust transactions by assisting in sending out more "trustworthy" information (Burt 1997, 2007), and in turn, help enhance CEOs' reputation and enforce CEOs' good behaviors. However, CEO social capital may release CEOs' concern on labor market and shield CEOs from internal and external governance, and therefore induces CEOs to participate in rent-seeking behaviors (e.g., El-Khatib et al., 2015). Overall, existing financial theories predict that CEO social capital can influence firm and market outcomes through the information and reputation channels.

Extant literature documents evidence to support that CEO social capital has positive or negative impacts on firm and market outcomes, such as mergers and acquisitions (El-Khatib et al., 2015), corporate risk-taking (Ferris et al., 2017), and capital investment (Fracassi, 2017). Moreover, a growing literature reveals that CEO social capital can influence a firm's overall information environment and financial reporting practices (Bhandari et al., 2018; Griffin et al., 2021), potentially including a firm's decision to use accruals in the financial reporting. As GAAP grants to managers flexibility to determine the use and magnitude of accruals, the use and magnitude of accruals reflects managerial incentives. Generally, there are two sets of managerial incentives: to provide private information to the market and to opportunistically manage earnings. We argue that CEO social capital can potentially influence CEOs' incentives, as well as their ability, to use discretionary accruals through the information and reputation channels.

Through the information channel, CEOs with higher social capital to learn more "soft" information from their network and decides whether, time, and how to use accruals. With the use of discretionary accruals, firms managed by CEOs with higher social capital can better communicate private information and signal firms' quality or managerial trust and competency to the market. The managed earnings, in turn, can help improve firm value by reducing the cost of equity (Francis et al., 2015). From an opposite perspective, CEO network itself represents less costly way to share information, firms managed by CEOs with higher social capital may not need to use accruals to achieve certain corporate goals.

Through the reputation channel, CEOs with higher social capital may direct the firms to use high-quality accruals that can better reflect future operating performance. The use of discretionary accruals, in turn, can help enhance the CEOs' reputation, and bring to the CEOs some additional benefits (e.g., increase of compensation). Inversely, if the use of discretionary accruals is of low-quality or even fraudulent, the practice could result in bad consequences to the firms and CEOs, such as the violation of GAAP, SEC litigation, and adverse media coverage. If CEOs care about their reputation, they may not want to direct the firms to use low-quality or fraudulent discretionary accruals. From a reverse perspective, CEO social capital may release CEOs from the labor market concerns (Liu, 2010) or shield CEOs from internal and external governance (El-Khatib et al., 2015), and therefore induce CEOs to engage in rent-seeking behaviors to benefit themselves. For instance, CEOs can direct the firms to opportunistically report discretionary accruals to meet or beat earnings targets, and potentially receive a higher incentive compensation.

It is also worth noting that extant literature demonstrates that managing earnings by accruals (sometimes referred to as the accruals-based earnings management) can be easily detected by SEC and firms might shift to managing earnings by the alteration of real operating activities (sometimes referred to as real earnings management), especially in post-SOX period (Cohen, Dey, and Lys, 2008; Cohen and Zarowin, 2010). As the use of discretionary accruals may have potential higher cost to the firms, managing earnings through discretionary accruals may represent a form of corporate risk-taking activities. As extant literature demonstrates that CEOs with higher social capital have the flexibility to tackle the risk-taking activities but can still achieve comparatively good outcomes (Ferris et al., 2017), we argue that firms managed by CEOs with higher social capital may be more likely to use discretionary accruals.⁸

⁸ Extant literature has long dissension on the potential cost of managing earnings through the alteration of real operating activities in the long run. One example is that firms that manage earnings upward through the alteration of real operating activities may be forced to skip valuable projects with positive NPV for purpose of managing earnings to meet or beat the target (Gunny, 2005; Griffen et al., 2021). However, the use of accruals, can eventually be reversed in the future. From this perspective, as long as the firms can use accruals in an appropriate manner, the use of accruals may not induce potential high cost of the firms in the long run.

Given the bright and dark side of CEO social capital on firms' choice of discretionary accruals, we hypothesize in the alternative format that

H1a: Firms managed by CEOs with higher social capital are more likely to use discretionary accruals to manage earnings upward.

H1b: Firms managed by CEOs with higher social capital are less likely to use discretionary accruals to manage earnings upward.

The null hypothesis is that there's no significant relation between CEO social capital and level of discretionary accruals used in the firms.

2.2. The use of income increasing (positive) vs. income decreasing (negative) discretionary accruals

Literature has long debate on the pros and cons for the use of income increasing (positive) or income decreasing (negative) discretionary accruals. A set of literature demonstrates that mangers have strong motivations to manage earnings upward to beat benchmark or analysts' estimate (Burgstahler and Dichev, 1997; Degeorge, Patel, and Zeckhauser, 1999). Additionally, Robin and Wu (2015) and Lee et al. (2006) find evidence to support that high-growth firms prefer to use income increasing discretionary accruals to signal favorable private information to external investors and to influence equity valuation, but those firms may incur a rising cost due to increasing amount of managed earnings. Further, Peasnell, Pope, and Young (2000) document evidence to suggest that firms with pre-managed earnings below zero or the prior year's earnings are more likely to use discretionary accruals to manage earnings upwards, while firms with pre-managed income well above these benchmarks may choose to adopt income decreasing discretionary accruals. Moreover, Ramanna and Roychodhury (2010) find that politically

connected firms with more extensive outsourcing activities are more likely to use income decreasing discretionary accruals to manage the political costs they face from negative outsourcing-related scrutiny. As the extant empirical evidence unveils both pros and cons for the use of income increasing and income decreasing discretionary accruals, we can not draw a conclusion on what specific discretionary accruals sub-sample CEOs with higher social capital may direct the firms to manage earnings upward. Thus, we hypothesize in the alternative format a dual hypothesis as follows:

H2a: If firms managed by CEOs with higher social capital use discretionary accruals to manage earnings upward, they prefer to use it in the income increasing discretionary (positive) accruals sub-sample.

H2b: If firms managed by CEOs with higher social capital use discretionary accruals to manage earnings upward, they prefer to use it in the income decreasing discretionary (negative) accruals sub-sample.

The null hypothesis is that firms managed by CEOs with higher social capital have no preference of managing earnings upward in income increasing or income decreasing discretionary accruals sub-sample.

2.3. CEO Social Capital and the pricing of discretionary accruals

As stated above, managers have incentives to direct the firms to use discretionary accruals to either provide private information to the market or opportunistically manage earnings. The information sent to the market allows the capital market to evaluate a firm's current performance, as well as predicting the firm's performance in the future, and hence to assign value to the firm (or saying price the firm). Such a signaling role links the use of discretionary accruals to the stock return of the firm.

The pricing of discretionary accruals has drawn great attention from researchers. Subramanyam (1996) is the first study that directly investigates the link by regressing returns on various components of earnings including discretionary accruals, and the study reports that market attaches value to discretionary accruals. However, although partially supporting the view that discretionary accruals improve the ability of earnings to reflect economic value, Subramanyam (1996) can not rule out the alternative view that discretionary accruals are opportunistic and value-irrelevant but priced by an inefficient market. Both Xie (2001) and Kraft, Leone, and Wasley (2007) report a negative association between discretionary accruals and future stock returns. Their findings imply that investors overprice discretionary accruals in the current period and thereby supports the alternative view from Subramanyam (1996). Robin and Wu (2015) demonstrate that managers tend to use discretionary accruals, especially the income increasing (positive) discretionary accruals, to signal future favorable performance while the capital market positively prices the discretionary accruals, especially the income increasing discretionary accruals. Hypothesizing that auditing plays an important role in mitigating aggressive and opportunistic reporting of discretionary accruals, Krishnan (2003) investigates the link between audit quality and the pricing of discretionary accruals and find that discretionary accruals used in the firms audited by "Big 6" auditors have greater associations with future profitability and stock return. The result indicates that market recognizes the superiority of good external governance mechanism in pricing of discretionary accruals.

As discussed earlier, CEO social capital can influence CEOs' incentives to use discretionary accruals through the information and reputation channel. We argue that CEO social

12

capital can potentially affect the pricing of discretionary accruals in the similar manner. If CEO social capital can motivate the CEOs to direct the firms to use high-quality discretionary accruals, and the capital market recognizes the situation, the discretionary accruals used in the firms managed by CEOs with higher social capital can better reflect the future operating performance and are thereby more relevant to firm value. Otherwise, the discretionary accruals used in those firms may be opportunistically reported and create more uncertainties for future operating situation, and thus would be priced negatively. It is also possible that capital market does not recognize the impact of CEO social capital on the use of discretionary accruals and does not assign positive or negative value for the process. The above discussion leads to our last two hypotheses, both in alternative format:

H3a: The association between the discretionary accruals and future operating performance is greater for firms managed by CEOs with higher social capital than for firms managed by CEOs with lower social.

H3b: The association between the discretionary accruals and future operating performance is weaker for firms managed by CEOs with higher social capital than for firms managed by CEOs with lower social.

H4a: The association between the discretionary accruals and stock return is greater for firms managed by CEOs with higher social capital than for firms managed by CEOs with lower social.

H4b: The association between the discretionary accruals and stock return is greater for firms managed by CEOs with higher social capital than for firms managed by CEOs with lower social. The null hypotheses for H3 and H4 are that there's no significant difference for the association between discretionary accruals and future operating performance and between discretionary accruals and stock return, respectively, for firms managed by CEOs with higher or lower social capital.

Section 3. Sample Construction, Variables Description, and Summary Statistics

3.1. Sample Construction

We construct our sample by collecting data from multiple datasets. To start with, we collect data from BoardEx dataset and construct CEO network centrality measurements to proxy for CEO social capital. The BoardEx dataset is a comprehensive data that includes personal information, educational histories, professional appointments, and firm information for hundreds of thousands of executives and non-executives at North American firms, non-profit organizations, and public entities. Specifically, we construct degree (*Degree*) and eigenvector centrality (*Eigenvector*) to capture the number of connections and the connection to the connected people and use principal component analysis to create a third centrality measurement, *PCA*, to capture the common features between Degree and Eigenvector. The detailed calculation process for CEO network centrality measurement is provided in section 3.2.

We next derive annual firm-level financial and return information from Compustat and CRSP, respectively, to calculate proxy for the use of discretionary accruals and other control variables. The detailed calculation process for the level of discretionary accruals used in a firm is provided in section 3.3.

In the final step, we merge the CEO network centrality measurements and the firm-level financials & return information to create our final sample for the empirical analysis. To enhance data homogeneity, we exclude financial (SIC 6000-6999) and utility firms (SIC 4900-4999) from our databases. We also delete observations with missing value for required variables in the empirical analysis. Our final sample consists of 12,953 firm-year observations from 1998 -2017. Out the sample, 8,833 and 4,102 firm-year observations report positive and negative discretionary accruals, respectively.

3.2. CEO Social Capital (Network Centrality) variables

Under the assumption that current relationships better capture the information advantage of the executives with higher social capital, we focus on current board appointments when calculating centrality measurements. To begin with, adding up the number of direct connections to other directors gives us degree centrality (*Degree*), which is simply the number of direct and current connections one has to other executives. *Degree* is hence a measure of the size of one's direct network. An extension of degree centrality is eigenvector centrality (*Eigenvector*), which takes into consideration the connected to other executives who, in turn, are connected to many other executives. *Eigenvector* is therefore an indication of the "importance" of one's network. Degree and eigenvector centralities are informative as to both the efficiency of information acquisition and dissemination given that executives with more (and better connected) connections are more likely to have access to more information at less cost, as well as with less reputational risk (since presumably more reputable and experienced executives are granted more board appointments). Specifically, we follow the extant literature (El-Khatib et al.,

2015; Egginton and McCumber, 2019; Egginton et al., 2022) and calculate the raw value for degree and eigenvector centrality measurements as:

$$Degree(D_i) = \Sigma_{ji} X_{ij} \tag{1}$$

where x_{ji} is 1 for presence of a social connection between i and j

$$Eigenvector (E_i) = satisfying \lambda E'E = E'AE$$
(2)

where E is an eigenvector of the matrix of connections A, and λ is its associated eigenvalue E_i is taken as the elements of the eigenvector E* associated with A's principal eigenvalue, λ^* .

Additionally, we also use principal component analysis to create a third centrality measurement, *PCA*, to capture the common features of degree and eigenvector centrality.

To construct the CEO network centrality measurements, we first create annual networks of U.S. executives from 1998-2017 and calculate raw value of degree and eigenvector centralities for all executives in the networks each year. While interpreting degree centrality results are intuitive, for example, 188 direct connections is likely "better" than 8 connections, eigenvector centrality results are not, as well as the PCA centrality. We thereby rank raw value of CEOs' network centrality into percentiles value each year such that, for example, a CEO in the 71st percentile of eigenvector centrality in a specific year has a higher eigenvector centrality than 70% of all executives in the network that same year.

3.3. Proxy for the level of discretionary accruals used in a firm

We follow extant literature (Subramanyam, 1996; Krishnan, 2003; Robin and Wu, 2015) and use cross-sectional modified Jones (1991) model to estimate the use of discretionary accruals. Specifically, by each year and industry (determined by 2-digit SIC code), we run the following cross-sectional regression for all the firms that are available in the Compustat dataset:

$$TlA_{cct} = \alpha + \beta_1 \times \left(\frac{1}{AT_{t-1}}\right) + \beta_2 \times (\Delta REV_t - \Delta RCT_t) + \beta_3 \times PP\&E_t + \varepsilon_t$$
(3)

where TlA_{cct} = total accruals of a firm at year t, calculated as difference between income before extraordinary items and operarting cash flow; AT_{t-1} = total asset at the end of year t-1; \triangle Rev_t = change in sales revenue for a firm at year t; \triangle Rec_t = change in receivables for a firm at year t; PP&E_t = gross PP&E for a firm at year t; All variables are scaled by lagged total assets.

The residual estimated from the above regression proxies for the level of discretionary accruals used in a firm. The higher values indicate higher-level discretionary accruals (can be either positive or negative) reported. We only keep the data with more than 15 observations by each year and industry.

We measure the level of discretionary accruals used in a firm using modified Jones (1991) model for several reasons: First, Lee et al. (2006) point out that controlling for performance and/or growth in the discretionary accruals model will lead to the reduction of the power of tests, as well as the underestimation of the amount of managed earnings for firms with better performance or higher growth. Second, several existing studies (Subramanyam, 1996; Bartov, Gul, and Tsui, 2000; DeFond and Subramanyam, 1998) evaluate the ability of accruals in detecting earnings management among several different models and find that the cross-sectional modified Jones (1991) model outperforms other time-series models. Finally, the use of the cross-

sectional modified Jones (1991) model allows us to compare our results with other related studies (e.g., Subramanyam, 1996).⁹

It is also worth noting that we also follow extant literature (Roychowdhury, 2006; Cohen et al., 2008; Zang, 2012) and create the proxy for firm's managing earnings with alternation of real operating activities (sometimes referred to as real earnings management) as one of the control variables. The detailed procedure to calculate the proxy is provided in appendix B.

3.4. Summary Statistics

Table 1 reports summary statistics. The mean (median) values for *Degree* and *Eigenvector* are 64 (63) and 57 (58), respectively. There is considerable heterogeneity in our sample of CEO social capital. For example, at the 25th percentile in sample *Degree* is in the 51st percentile of all executives in the network while at the 75th percentile *Degree* is in the 78th percentile of all executives; the standard deviation is approximately 17.34 percentiles.

One other key variable, *Dis Accruals*, has a mean (median) value of 0.06 (0.05), with about 68% of the values being positive. While the value of *Dis Accruals* is left-skewed, we do observe approximately equivalent 1^{st} (Q1) and 3^{rd} (Q3) quartiles relative to the median value, suggesting a broadly symmetric distribution of *Dis Accruals* around the median value.

[Insert Table 1 about here]

4. Research Methodology and Empirical Results

4.1. CEO Social Capital and the use of Discretionary Accruals

⁹ As we can not completely rule out the possibility that the potential misspecification problem of the modified Jones (1991) model drives our result, in an un-tabulated analysis, we use discretionary accruals estimated by the performance-matched modified Jones (1991) model (introduced by Kothari, Leone, and Wasley, 2005) in all our regressions and find similar result.

4.1.1. CEO Social Capital and the Level of Discretionary Accruals used in a firm

To test our first hypothesis, we regress the level of discretionary accruals used in a firm (*Dis Accruals*) on CEO social capital measurements and other control variables (including the level of real earnings management used in the firm) in a pooled OLS regression as:

 $Dis Accruals_{i,t} = \alpha + \beta_{i,t}CEO Social Capital + \delta_{i,t}Controls + \lambda_{i,t} Industry + \gamma_{i,t}Year + \mathcal{E}_{i,t}$ (4)

The dependent variable is the level of discretionary accruals used in a firm (Dis Accruals). Our variable-of-interests are the CEO social capital measurements. To separate the effect of CEO social capital on the use of discretionary accruals from other factors, we include firm-level and CEO characteristics that are found to affect the use of discretionary accruals (Dechow, Ge, and Schrand., 2010; Dechow, Ge, Larson, and Sloan., 2011, Dechow and Dichev. 2002; Roychowdhury 2006; Griffin et al. 2021; Zang, 2012; Chan et al., 2015). Specifically, we include firm size (Size), leverage (Leverage) and return on asset (ROA), to control for scale effect and profitability of a firm (Kothari et al., 2005; Cohen et al., 2008). As Barth, Beaver, Hand, and Landsman. (1999), Skinner and Sloan (2002), and Zang (2012) find that the incentives to report higher earnings figure increase with firms' growth potential, we thereby include Book-to-Market equity ratio (BTM) to control for firms' growth potential. We also control for firms' volatility of profitability (ROAstd) and cash flow (CFstd) because firms are more likely to use discretionary accruals to manage earnings in volatile performance environment. Following Zang (2012) and Chan et al. (2015), we include Growth of sale (Sales Growth) and use of big four accounting firms as auditor (*Big4*) to control for cost of managing earnings using discretionary accruals. Dechow and Dichev (2002) suggest that longer operating cycles indicate more operational uncertainty that may possibly lead to use of earnings management, so we also include firms'

operating cycle (*Cycle*) in the regression. Following Ali and Zhang (2015), we control for other CEO characteristics such as numbers of years a CEO is in the position (*Tenure*), and whether the CEO simultaneously serves as board director (*Duality*). Finally, we incorporate in the regression the level of real earnings management used in a firm to capture the substitution effect between managing earnings with the use of discretionary accruals and the alteration of real operating activities (Cohen et al., 2008; Zang, 2012). All regressions include time and industry (determined by the classifications of 48 industries from Fama and French, 1997) fixed effects, and errors are robust to firm heteroscedasticity. T-stats are reported in parentheses.

Table 2 reports the results of pooled OLS regressions of CEO social capital, on the level of discretionary accruals used in a firm. As can be seen from the table, the coefficients for all CEO social capital measurements are positive and significant (p<0.1). Set *Degree* (column (1)) as an example to illustrate the marginal effect: Holding other constant, on average one unit (equal to 1%) increase in degree centrality will lead to increasing use of discretionary accruals that equals to about 0.0423% of asset. The result is also economically significant, for instance, moving from 25^{th} to 75^{th} percentile degree centrality in the sample, the level of discretionary accruals used increase by around 19% of the mean value of discretionary accruals in the sample.¹⁰ Of all the control variables, *Real EM, Size, ROA, Leverage, ROAstd*, and *CFstd* show significant statistical relation (either positive or negative) with the dependent variable. Overall, the results from table 2 illustrate that, holding other constant, CEO social capital has a significantly positive correlation with the level of discretionary accruals used in a firm.

Motivated by the Griffin et al. (2021), we re-run equation (4) by switching the places of *Dis Accruals* (as the independent variable) and *Real EM* (as one of the dependent variables).

 $^{^{10}}$ 19% is calculated by 0.000423 \times (78 – 51) / 0.06

Table 3 reports the results. As can be seen from the table, the coefficients for all CEO social capital measurements are negative and highly significant (p<0.01). The results from table 3 reveal that, holding other constant, CEOs social capital has a significantly negative correlation with the level of real earnings management in a firm.¹¹ Combined the results from tables 2 and 3 together, we conclude that firms managed by CEOs with higher social capital are more likely to manage earnings upward using discretionary accruals, rather than alter real operating activities. Overall, the results from this section support H1a.

[Insert Tables 2 and 3 about here]

4.1.2. Income Increasing (Positive) vs. Income Decreasing (Negative) Discretionary Accruals

To test our second hypothesis, we divide our sample into two sub-samples based on whether a firm uses income increasing (positive) or income decreasing (negative) discretionary accruals and re-run equation (4) in each sub-sample. Table 4 reports the results. As can be seen from the table, the coefficients for CEO social capital in columns (1), (3), and (5) are positive but insignificant, whereas the ones in columns (2), (4), and (6) are positive and highly significant (p<0.05). The result indicates that the significantly positive association between CEO social capital and the level of discretionary accruals used in a firm concentrates in the income decreasing (negative) discretionary accruals sub-sample. Stated otherwise, compared with firms managed by CEOs with lower social capital, firms managed by CEOs with higher social capital are more likely to manage earnings upward in the income decreasing (negative) discretionary accruals sub-sample. It is also worth noting that the table also reveals a significantly negative

¹¹ Further analyses (result un-tabulated) show that the significantly negative relation between the real earnings management and CEO social capital measurements comes mainly from the effects of abnormal production cost and abnormal discretionary expenditure. Please refer to the appendix for the definition and the calculation process for the components of the real earnings management.

coefficient and a significantly positive coefficient for *BTM* and *Sales Growth* in columns (1), (3), and (5). The result is consistent with the finding from extant literature that high-growth firms are more likely to report positive discretionary accruals to potentially signal positive private information to the market and to influence equity valuation (Robin and Wu, 2015; Lee et al., 2006). Overall, the results from this section support H2a.

[Insert Table 4 about here]

4.1.3. Robustness Check

To ensure that the results from the previous sections are not biased due to the use of OLS regression model, the self-selections bias, the omitted variables, and the endogeneity issue, we apply the following robustness analyses.

4.1.3.1. Two-stage IV regression

It may be that firms that more likely to manage earnings upward using discretionary accruals hire CEOs with higher social capital and reputable. In turn, we attempt to triangulate against possible reverse causality and endogeneity concerns by applying the two-stage instrumental variable (IV) regression. Under the assumption that county mean CEO social capital will be correlated with firm CEO social capital but should not be a determinant of the level of discretionary accruals used in a firm, we instrument CEO social capital with county mean CEO social capital and rerun the analyses. Table 5 presents the result for the second stage of the twostage instrumental variable (IV) regression. As indicated in the table, the coefficients for the CEO social capital measurements remain positive and highly significant in columns (3), (6), and (9). The result from table 5 reinforce the finding from table 4 that firms managed by CEOs with higher social capital are more likely to manage earnings upward in the income decreasing (negative) discretionary accruals sub-sample.

[Insert Table 5 about here]

4.1.3.2. Difference-in-difference analysis: Evidence from CEO turnover

To further mitigate endogeneity concerns, we use CEO turnover as an experiment and apply a difference-in-difference analysis. We admit that CEO turnover events may not be exogenous in nature as it could be affected by both CEO social capital and the use of discretionary accruals in a firm, so this test cannot completely resolve the endogeneity issue, instead, it serves as additional evidence to reinforce the finding from the baseline regression. Specifically, we focus on comparing the difference between two types of CEO turnover made by firms that were initially managed by CEOs with lower social capital. For the first type, firms switch to new CEOs with higher social capital measured by Degree, Eigenvector, and PCA (the treatment group). For the second type, firms literally switch to new CEOs with lower social capital (the control group). We use indicator variables (treat_CEO Social Capital) to define whether a firm belongs to the treatment group, or the control group; specially, the indicator variables (Treat CEO Social Capital) include Treat Degree, Treat Eigenvector, and Treat PCA based on the CEO social capital value measured by *Degree*, *Eigenvector*, and *PCA*, respectively. The indicator variables equal one if a firm has an average CEO social capital value below the median value each year for the pre-transition period and an average CEO social capital value above the median value each year for the post-transition period, and zero if a firm has an average CEO social capital value below the median value for both the pre-transition and post-transaction periods. We use the indicator variable, *Post*, to represent observations following CEO turnover. We estimate the following regression:

 $Dis Accruals_{i,t} = \alpha + \beta_{1,i,t} Treat_CEO Social Capital + \beta_{2,i,t} Post + \beta_{1,i,t} Post + \beta_{2,i,t} Post + \beta_{2,i,t}$

 $\beta_{3,i,t}$ Treat_CEO Social Capital × Post + $\delta_{i,t}$ Controls + $\lambda_{i,t}$ Industry + $\gamma_{i,t}$ Year + $\varepsilon_{i,t}$ (5)

Our variable of interest is the coefficient $(\beta_{3,i,t})$ for the interaction term between the two dummy variables (*Treat_CEO social capital* × *Post*). We expect $\beta_{3,i,t}$ to be positive and significant if an increase of CEO social capital following the CEO turnover results in higher level of discretionary accruals used in a firm.

We construct our CEO turnover samples by identifying the change of DirectorID in a firm within our sample. Additionally, we require the CEO turnover sample to fulfill the following criteria: (1) The CEOs in both pre- and post-transition period must work for two consecutive years (excluding the transition year); (2) To avoid the confounding effect of multiple CEO turnovers on our results, if a firm changes its CEO more than once, we only count the first CEO turnover and ignore the subsequent CEO turnover in our sample period. Defining our sample in this manner, our treatment groups consist of 9, 20, and 13 firms, whereas our control group consist of 30, 42, and 37 firms based on the CEO social capital value measured by *Degree*, *Eigenvector*, and *PCA*, respectively.

Table 6 reports the results for the difference-in-differences analysis. As can be seen from the table, the coefficients for *Treat_CEO social capital* × *Post* in columns (3) and (9) are positive and significant (p<0.1) while the one in column (6) are positive but insignificant. The evidence suggests that compared with firms that switch from CEOs with lower social capital to CEOs with lower social capital, firms that switch from CEOs with lower social capital to CEOs with higher social capital significantly increase the level of discretionary accruals in the income decreasing

(negative) discretionary accruals sub-sample. Similar to the one from table 5, the result from table 6 reinforces the findings from table 4.¹²

[Insert Table 6 about here]

4.1.3.3. Propensity score matching

In an un-tabulated analysis, we create dummy variables to proxy for the high (if equal to 1) and low (if equal to 0) CEO social capital based on the median value of each CEO social capital measurement, replace the dummy variables with the percentile value of the CEO social capital measurements into the regression, and find similar results (un-tabulated). Furthermore, as the systematic differences between firms managed by CEOs with high versus low social capital firms may influence our results in the previous sections, we employ a propensity score method (closest neighbor method) to alleviate the endogeneity concern (including the self-selection bias). ¹³ Specifically, we first run equation (4) without the CEO social capital measurements and predict the level of discretionary accruals; we next rank the predicted level of discretionary accruals by year and industry (measured by 2-digit SIC code); moreover, we use propensity score method (closest neighbor method) to match firms managed by CEOs with higher with the ones managed by CEOs with lower social capital to create a reduced sample; finally, we re-run equation (4) including the percentile value of the CEO social capital measurements within the reduced sample predicted by the propensity score matching method. Table 7 reports the result. As can be seen from the table, the coefficients for the CEO social capital measurements in

¹² Although the coefficient for *Treat_Eigenvector* \times *Post* in column (6) is positive but not significant, we argue that the result from the difference-in-difference analysis in table 6 still supports the findings from table 4 as the coefficient for *Treat_PCA* \times *Post* in column (9) is positive and marginally significant.

¹³ In an un-tabulated analysis, we also apply a new method with similar function to the propensity score matching, called entropy balancing. The use of entropy balancing allows for the retention of all sample firms but assigns different weights for different observations in the regression. The results remain identical after applying the entropy balancing method.

columns (1), (4), and (7), as well as in columns (3), (6), and (9), are positive and significant (p<0.1). The result from table 7 confirms the findings from tables 2 and 4 that firms managed by CEOs with higher social capital are more likely to manage earnings upward by discretionary accruals, particularly in the income decreasing (negative) discretionary accruals sub-sample. It is also worth noting that the results remain the same (un-tabulated) if we include the dummy variables (of high and low CEO social capital), rather than the percentile value of the CEO social capital measurements, in the regression within the reduced sample.

[Insert Table 7 about here]

4.1.3.4. Controlling for managerial ability

Extant literature reports that managerial ability may affect the use of discretionary accruals (Demerjian, Lev, Lewis, and McVay, 2013), so managerial ability represents the potential omitted variable for the regression. To ensure that our findings are primarily driven by the effect of social capital, rather than solely from the effect of human capital of management team in a firm, we control for managerial ability (Demerjian et al. 2013) in equation (4), and we still find similar results (un-tabulated).

4.1.4. CEO Social Capital and the use of Discretionary Accruals: Information vs Reputation Channels (Effects)

CEO social capital measures may proxy for both reputation and information effects. In this section, we analyze which effect(s) drive the use of discretionary accruals. To do so, we follow the literature (e.g., Egginton, McBrayer, and McCumber., 2022) to create proxies for information and reputation channels (effects). Specifically, we first regress the aggregate CEO social capital measurements on a vector of plausible variables of firm and personal CEO characteristics that are likely to be correlated with our social capital measurements. Since larger and more profitable firms are more likely to hire reputable and experienced CEOs, the vectors of firm characteristics contain firm size, leverage, and profitability. The vectors of CEO personal characteristics include the total executive tenure of the executive (from first executive appointment at any firm), the total number of boards the CEOs have served, whether the CEO has a degree from an elite university, and whether and how many times the CEOs has been recognized with a meaningful award.¹⁴ The predicted value and the residual from this stage are proxies for the reputation and information channels (effects), respectively. Next, we substitute the proxies for reputation and information channels (effects) into equation (4) for the CEO social capital measurements and rerun the regression. Table 8 reports the result. As can be seen from columns (1) and (3), the coefficients for *Information Channel* are positive and significant (p<0.1). Similarly, columns (4) and (6) also report positive and significant coefficients for *Reputation Channel*. The overall results suggests that both information and reputation effects contribute to the level of discretionary accruals used in a firm.

[Insert Table 8 about here]

4.2. CEO Social Capital and the Pricing of Discretionary Accruals

4.2.1. CEO Social Capital, the use of Discretionary Accruals, and the Future Operating Performance of a firm

To test our third hypothesis, we follow the extant literature (Subramanyam, 1996; Krishnan, 2003; Kraft et al., 2007; Robin and Wu, 2015) and run the following regression:

¹⁴ "Elite" university is determined by creating a list of the top 100 global institutions according to *The Financial Times*, *US News and World Report*, and *The Times Higher Education* world university rankings. These institutions include Ivy League universities, Stanford, Chicago, London School of Business, etc. "Meaningful" awards are those parsed from BoardEx data and include regional, national, or international recognitions such as *Forbes*' "Top 20 CEOs."

Future Operating Performance_{*i*,*t*+1} = $\alpha + \beta_{1,i,t}$ Opearting CF + $\beta_{2,i,t}$ NA Accruals + $\beta_{3,i,t}$ Dis Accruals + $\beta_{4,i,t}$ High CEO Social Capital + $\beta_{5,i,t}$ Operating CF × High CEO Social Capital + $\beta_{6,i,t}$ NA Accruals × High CEO Social Capital + $\beta_{7,i,t}$ DIS Accruals × High CEO Social Capital + $\delta_{i,t}$ Controls + $\lambda_{i,t}$ Industry + $\gamma_{i,t}$ Year + $\varepsilon_{i,t}$ (6)

where *Future Operating Performance* is presented by either the one-year ahead return on asset (*Future ROA*), or the probability of dividend increase one-year ahead (*Future DV Increase*); Operating CF, NA Accruals, and Dis Accruals represent current cash flow from operation, the level of non-discretionary accruals, and the level of discretionary accruals;¹⁵ Controls is a vector of control variables common in extant literature.¹⁶ The variables-of-interest are the interaction terms between CEO social capital and level of discretionary accruals used in a firm (*High CEO Social Capital × Dis Accruals*).¹⁷

Table 9 and 10 present the results, with dependent variable as *Future ROA* and *Future DV Increase*,¹⁸ respectively. As can be seen from table 9, the coefficients for *Dis Accruals* are positive and significant in columns (1), (2), (4), (5), (7), and (8), supporting the finding from extant literature that the use of discretionary accruals, especially the positive ones, can have positive predictability on future profitability of a firm (Dechow, 1994; Robin and Wu, 2015; Lee

¹⁵ The three variables represent the decomposition of the current earnings. All the variables are scaled over lagged total asset.

¹⁶ Detailed descriptions of all variables, including control variables, are provided in Appendix A.

¹⁷ To avoid the potential multicollinearity issue caused by the interaction between the percentile value of CEO social capital and the level of discretionary accruals used in a firm, we use the dummy variable of high- and low- CEO social capital group, rather than the percentile value of CEO social capital, in the regression. Additionally, using dummy variable allows us to see the economic significance for the impact of CEO social capital on the association between the level of discretionary accruals and the future operating performance. Similar situation applies in equation (7).

¹⁸ We use logistic regression in table 9 because the dependent variable is an indicator variable, whereas we use OLS regression in table 8 because the dependent variable is a continuous variable.

et al., 2006). The coefficients for *High CEO Social Capital* × *Dis Accruals* are positive and significant (p<0.05) in columns (1), (3), (4), (6), (7), and (9), indicating that the CEO social capital has a significantly positive impact on the predictability of discretionary accruals, especially the negative ones, towards the future profitability of a firm.¹⁹ Similar situation can be observed in table 10 except that the coefficients for *High CEO Social Capital* × *Dis Accruals* are positive and significant (p<0.1) only in columns (3), (6), and (9), but are positive and insignificant in columns (1), (4), and (7).

Overall, the results from tables 9 and 10 proves that compared with the ones used in the firms managed by CEOs with lower social capital, the discretionary accruals used in the firms managed by CEOs with higher social capital, particularly the negative ones, can better predict firms' future operating performance, and thus supports H3a.

[Insert Tables 9 and 10 about here]

4.2.2. CEO Social Capital, the use of Discretionary Accruals, and the Future Stock Return of a firm

As can be concluded from the previous sub-section, the association between negative discretionary accruals and future operating performance is greater for firms managed by CEOs with higher social capital, it is thereby reasonable to test whether the capital market recognizes the situation and assign positive values for those firms.²⁰ To test the hypothesis, we follow the extant literature (Subramanyam, 1996; Krishnan, 2003; Kraft et al., 2007; Robin and Wu, 2015) and run the following regression:

¹⁹ In an un-tabulated analysis, we also re-run the equation (6) using the future operating cash flow to proxy for the future profitability of a firm, but only find that CEO social capital has a positive but insignificant impact on the predictability of discretionary accruals towards the future operating cash flow.

²⁰ This is to formally test the pricing of the discretionary accruals (by capital market), or to test our 4th hypothesis.

Stock Return_{*i*,*t*+1} = α + $\beta_{1,i,t}$ Opearting CF + $\beta_{2,i,t}$ NA Accruals + $\beta_{3,i,t}$ Dis Accruals + $\beta_{4,i,t}$ High CEO Social Capital + $\beta_{5,i,t}$ Operating CF × High CEO Social Capital + $\beta_{6,i,t}$ NA Accruals × High CEO Social Capital + $\beta_{7,i,t}$ DIS Accruals × High CEO Social Capital + $\delta_{i,t}$ Controls + $\lambda_{i,t}$ Industry + $\gamma_{i,t}$ Year + $\varepsilon_{i,t}$ (7)

where *Stock Return* is the compounded monthly stock returns for a twelve-month period ending three months after the end of the fiscal year of the firm; Other variables are the same as the ones used in equation (6).

Table 11 reports the results. As can be seen from the table, the coefficients for *Dis Accruals* are positive and significant in columns (2), (5), and (8), but are negative and significant in columns (3), (6), and (9), indicating that the capital market positively prices the level of income increasing (positive) but negatively price the income decreasing (negative) discretionary accruals used in the firms managed by CEOs with lower social capital. Interestingly, the coefficients for *High CEO Social Capital* × *Dis Accruals* are positive and significant (p<0.1) in columns (3), (6), and (9), demonstrating that the CEO social capital has significantly positive impact on the pricing of the negative discretionary accruals (by the capital market).

Thus, the result from table 11 supports H4a that capital market prices the (negative) discretionary accruals used in the firms managed by CEOs with higher social capital in a more positive manner than the ones used in the firms managed by CEOs with lower social capital.²¹

[Insert Table 11 about here]

Section 5: Conclusion

²¹ It is worth noting that he results from tables 9, 10, and 11 are robust after controlling for the proxy for the managerial ability from Demerjian et al. (2013).

Our study investigates the relationship between CEO social capital and the use of discretionary accruals in a firm, as well as the pricing of discretionary accruals from the capital market. Our analyses reveal that firms managed by CEOs with higher social capital are more likely to use discretionary accruals (predicted by cross-sectional Modified Jones (1991) Model) to manage earnings upward, rather than alter firms' real operating activities. However, those behaviors concentrate in the income decreasing (negative) discretionary accruals sub-sample. The result is robust with the control for self-selections bias, the omitted variables, and the endogeneity. Further analysis unveils that compared with the ones used in the firms managed by CEOs with higher social capital, the discretionary accruals used in the firms managed by CEOs with higher social capital, especially the ones in the income decreasing (negative) discretionary accruals used accruals sub-sample, can better reflect the future operating performance and are thus more relevant to the future stock return of the firms.

As extant literature documents evidence to suggest that the use of discretionary accruals may incur higher cost for the firm (Cohen et al., 2008; Cohen and Zarowin, 2010), managing earnings through discretionary accruals may thereby represents a form of corporate risk-taking activities. Moreover, as indicated in the previous studies, the use of negative discretionary accruals may not reflect the future operating earnings, and thus, is priced negatively by the capital market. Our findings regarding the more frequent use of discretionary accruals to manage earnings upward, particularly in the income decreasing (negative) discretionary accruals subsample, thus reveal that CEOs with higher social capital are more likely to engage in risk-taking activities, consistent with the findings from previous literature (Faleye et al., 2014; Ferris et al., 2017). However, as CEOs with higher social capital have the degree of freedom to operate (Diamond, 1985), their risk-taking won't cause bad consequences. To some extent, such risktaking can benefit the firm by improving its overall informational environment through the supply of appropriate level of negative discretionary accruals that can reflect the firms' future operating performance, and thus are positively priced by the capital market.

References

- Ali, A., & Zhang, W. (2015). CEO tenure and earnings management. *Journal of Accounting and Economics*, *59*(1), 60-79.
- Barth, M. E., Beaver, W. H., Hand, J. R., & Landsman, W. R. (1999). Accruals, cash flows, and equity values. *Review of Accounting Studies*, 4(3), 205-229.
- Bartov, E., Gul, F. A., & Tsui, J. S. (2000). Discretionary-accruals models and audit qualifications. *Journal of Accounting and Economics*, *30*(3), 421-452.
- Becker, C. L., DeFond, M. L., Jiambalvo, J., & Subramanyam, K. R. (1998). The effect of audit quality on earnings management. *Contemporary Accounting Research*, 15(1), 1-24.
- Bhandari, A., Mammadov, B., Shelton, A., & Thevenot, M. (2018). It is not only what you know, it is also who you know: CEO Network Connections and Financial Reporting Quality. *Auditing: A Journal of Practice & Theory*, *37*(2), 27-50.
- Black, E.L., Christensen, T.E., Joo, T.T., & Schmardebeck, R. (2017b). The relation between earnings management and non-GAAP reporting. *Contemporary Accounting Research*, *34*(2), 750-782.
- Bowen, R. M., Burgstahler, D., & Daley, L. A. (1987). The incremental information content of accrual versus cash flows. *Accounting Review*, 723-747.
- Burgstahler, D., & Dichev, I. (1997). Earnings management to avoid earnings decreases and losses. *Journal of Accounting and Economics*, 24(1), 99-126.
- Burt, Ronald S. (1997). The Contingent Value of Social Capital. *Administrative Science Quarterly*, 42(2), 339-365.
- Burt, R. S. (2007). Brokerage and closure: An introduction to social capital. OUP Oxford.
- Chan, L. H., Chen, K. C., Chen, T. Y., & Yu, Y. (2015). Substitution between real and accrualsbased earnings management after voluntary adoption of compensation clawback provisions. *The Accounting Review*, *90*(1), 147-174.
- Cohen, D. A., Dey, A., & Lys, T. Z. (2008). Real and accrual based earnings management in the pre- and post- Sarbanes-Oxley periods. *The Accounting Review*, 83(3), 757-787.
- Cohen, D. A., & Zarowin, P. (2010). Accrual-based and real earnings management activities around seasoned equity offerings. *Journal of Accounting and Economics*, 50(1), 2-19.
- Dechow, P. (1994). Accounting earnings and cash flows as measures of firm performance: The role of accounting accruals. *Journal of Accounting and Economics*, *17*, 3-42.
- Dechow, P. M., & Dichev, I. D. (2002). The quality of accruals and earnings: The role of accrual estimation errors. *The Accounting Review*, 77(s-1), 35-59.

- Dechow, P. M., Ge, W., Larson, C. R., & Sloan, R. G. (2011). Predicting material accounting misstatements. *Contemporary Accounting Research*, 28(1), 17-82.
- Dechow, P., Ge, W., & Schrand, C. (2010). Understanding earnings quality: A review of the proxies, their determinants and their consequences. *Journal of Accounting and Economics*, 50(2-3), 344-401.
- Dechow, P. M., Kothari, S. P., & Watts, R. L. (1998). The relation between earnings and cash flows. *Journal of Accounting and Economics*, 25(2), 133-168.
- Dechow, P. M., Sloan, R. G., & Sweeney, A. P. (1995). Detecting earnings management. *Accounting Review*, 193-225.
- Degeorge, F., Patel, J., & Zeckhauser, R. (1999). Earnings management to exceed thresholds. *The Journal of Business*, 72(1), 1-33.
- Demerjian, P. R., Lev, B., Lewis, M. F., & McVay, S. E. (2013). Managerial ability and earnings quality. *The Accounting Review*, 88(2), 463-498.
- Diamond, D. W. (1985). Optimal release of information by firms. *The Journal of Finance*, 40(4), 1071-1094.
- Egginton, J. F., and McCumber, W. R. (2019). Executive Network Centrality and Stock Liquidity. *Financial Management*, *48*(3), 849-871.
- Egginton, J. F., McBrayer, G., & McCumber, W.R. (2022). Executive Networks and Global Stock Liquidity. *Journal of Financial Research*, forthcoming.
- El-Khatib, Rwan., Fogel, Kathy., & Jandik, Tomas. (2015). CEO network centrality and merger performance. *Journal of Financial Economics*, *116*(2), 340-382.
- Faleye, Olubunmi., Kovacs, Tunde., & Venkateswaran, Anand. (2014). Do better-connected CEOs innovate more? *Journal of Financial and Quantitative Analysis*, 49(5-6), 2101-1225.
- Fama, E.F. (1980). Agency problems and the theory of the firm. *Journal of Political Economy*, 88(2).
- Fama, Eugene F., and French, Kenneth R. (1997). Industry costs of equity. *Journal of Financial Economics*, *43*(2), 153-193.
- Ferris, S.P., Javakhadze, D., & Rajkovic, T. (2017a). CEO social capital, risk-taking and corporate policies. *Journal of Corporate Finance*, 47, 46-71.
- Fracassi, Cesare. (2017). Corporate Finance Policies and Social Networks. *Management Science*, 63(6), 2397-2771.
- Francis, J., LaFond, R., Olsson, P., & Schipper, K. (2005). The market pricing of accruals quality. *Journal of Accounting and Economics*, *39*(2), 295-327.

- Griffin, Paul A., Hong, Hyun A., Liu, Yun., & Ryou, Ji Woo. (2021). The dark side of CEO social capital: Evidence from real earnings management and future operating performance. *Journal of Corporate Finance*, 68, 101920.
- Guay, W. R., Kothari, S. P., & Watts, R. L. (1996). A market-based evaluation of discretionary accrual models. *Journal of Accounting Research*, *34*, 83-105.
- Gunny, K. A. (2005). What are the consequences of real earnings management? University of California, Berkeley.
- Hambrick, Donald C., and Mason, Phyllis A. (1984). Upper Echelons: The organization as a reflection of its top managers. *The Academy of Management Review*, 9(2),193-206.
- Healy, P. M., & Palepu, K. G. (1993). The effect of firms' financial disclosure strategies on stock prices. *Accounting horizons*, 7(1), 1.
- Holthausen, R. W. (1990). Accounting method choice: Opportunistic behavior, efficient contracting, and information perspectives. *Journal of Accounting and Economics*, 12(1-3), 207-218.
- Jones, J. (1991). Earnings management during import relief investigations. *Journal of Accounting Research*, 29, 193-228.
- Kothari, S. P., Leone, A. J., & Wasley, C. E. (2005). Performance matched discretionary accrual measures. *Journal of Accounting and Economics*, *39*(1), 163-197.
- Kraft, A., Leone, A. J., & Wasley, C. E. (2007). Regression-based tests of the market pricing of accounting numbers: The Mishkin test and ordinary least squares. *Journal of Accounting Research*, 45(5), 1081-1114.
- Krishnan, G. V. (2003). Does Big 6 auditor industry expertise constrain earnings management? *Accounting horizons*, 17, 1-16.
- Lee, C. W. J., Li, L. Y., & Yue, H. (2006). Performance, growth and earnings management. *Review of Accounting Studies*, *11*(2), 305-334.
- Liu, Y. (2010). The role of networks in the CEO and director labor market. University of Maryland, College Park.
- Nohria, N. (1992). Is a network perspective a useful way of studying organizations? N. Nohria, RG Eccles, eds. *Networks and Organizations*, 1-22.
- Peasnell, K. V., Pope, P. F., & Young, S. (2000). Accrual management to meet earnings targets: UK evidence pre-and post-Cadbury. *The British Accounting Review*, *32*(4), 415-445.
- Ramanna, K., & Roychowdhury, S. (2010). Elections and discretionary accruals: Evidence from 2004. *Journal of Accounting Research*, *48*(2), 445-475.
- Robin, A., & Wu, Q. (2015). Firm growth and the pricing of discretionary accruals. *Review of Quantitative Finance and Accounting*, 45(3), 561-590.

- Roychowdhury, S. (2006). Earnings management through real activities manipulation. *Journal of Accounting and Economics*, 42(3), 335-370.
- Skinner, D. J., & Sloan, R. G. (2002). Earnings surprises, growth expectations, and stock returns or don't let an earnings torpedo sink your portfolio. *Review of Accounting Studies*, 7(2), 289-312.
- Subramanyam, K. R. (1996). The pricing of discretionary accruals. *Journal of Accounting and Economics*, 22(1-3), 249-281.
- Watts, R. L., & Zimmerman, J. L. (1990). Positive accounting theory: A ten year perspective. *Accounting Review*, 131-156.
- Woolcock, Michael. (1998). Social capital and economic development: Toward a theoretical synthesis and policy framework. *Theory and Society*, 27, 151–208.
- Xie, H. (2001). The mispricing of abnormal accruals. *The Accounting Review*, 76(3), 357-373.
- Zang, A. Y. (2012). Evidence on the trade-off between real activities manipulation and accrualbased earnings management. *The Accounting Review*, 87(2), 675-703.

Appendix A: Variables Descriptions

The Appendix documents the definitions of variables used in this study ("_w" indicates that the value of the variables is winsorized at 1% and 99% level).

Variables	Descriptions
Dependent Variables:	
Dis Accruals	The error term estimated by the cross-sectional Modified Jones (1991) Model
Real EM	The sum of abnormal prediction cost, abnormal cash flow and abnormal discretionary expense, suggested by Roychowdhury (2006) Model
Future ROA	One-year ahead return on asset
Future DV Increase	Indicator variable that equals one if dividend payout of a firm increases one-year ahead, and zero otherwise.
Stock Return	The cumulative stock return for the twelve-month period ending three months after the fiscal year end
Variables-of-interest:	
Degree	Number of direct ties with others in the network
Eigenvector	Importance of an individual in the network
PCA	Principal Component Analysis of the percentile value of Degree and Eigenvector
Instrumented Degree	The predicted value for degree centrality by regressing a firm's CEO degree centrality on the instruments of the county mean CEO degree centrality (without the firm), along with the control variables
Instrumented Eigenvector	The predicted value for eigenvector centrality by regressing a firm's CEO eigenvector centrality on the instruments of the county mean CEO eigenvector centrality (without the firm), along with the control variables
Instrumented PCA	The predicted value for PCA centrality by regressing a firm's CEO PCA centrality on the instruments of the county mean CEO PCA centrality (without the firm), along with the control variables
Information Channel	The residual value generated by regressing the raw aggregate CEO network centrality data on a vector of plausible variables of firm and personal CEO characteristics that are likely to be correlated with the CEO network centrality measures
Reputation Channel	The predicted value generated by regressing the raw aggregate CEO network centrality data on a vector of plausible variables of firm and personal CEO characteristics that are likely to be correlated with the CEO network centrality measures.
Treat_Degree	Indicator variable that equals one if a firm has an average value of degree below the median value each year for the pre-transition period and an average value of degree above the median value

	each year for the post-transition period, and zero if a firm has an
	average value of degree below the median value for both the pre-
	transition and post-transaction periods.
Treat_Eigenvector	Indicator variable that equals one if a firm has an average value
	of eigenvector below the median value each year for the pre-
	transition period and an average value of eigenvector above the
	median value each year for the post-transition period, and zero if
	a firm has an average value of eigenvector below the median
	value for both the pre-transition and post-transaction periods.
Treat_PCA	Indicator variable that equals one if a firm has an average value
	of PCA below the median value each year for the pre-transition
	period and an average value of PCA above the median value
	each year for the post-transition period, and zero if a firm has an
	average value of PCA below the median value for both the pre-
	transition and post-transaction periods.
Post	Indicator variable that equals one for post-CEO transition, and
	zero for pre-CEO transition
Operating CF	Cash flow from operation of a firm in year t.
NA Accruals	The predicted value estimated by the cross-sectional Modified
	Jones (1991) Model; or the difference between total accruals and
	Dis Accruals
Controlled Variables:	
Size	Firm size at year t (natural log of assets)
BTM	Book to Market equity ratio at year t; calculated as book value
	divided by market value of assets, of which market value of total
	assets is equal to book value of total assets minus common
	equity plus market value of equity
ROA	Return on assets at year t
Leverage	Total debt scaled by total asset at year t
ROAstd	Rolling standard deviation of return on assets for the past three
	years including the current year
CFstd	Rolling standard deviation of operating cash flow on assets for
	past three years including current year
Cycle	Thousand days' receivable plus the days' inventory less days'
	payable (operating cycle divided by one thousand)
Sales Growth	Sales growth in year t
Big4	Indicator variable that equals one if a firm is audited by big four
	auditors, zero otherwise
Duality	Indicator variable that equals one if CEO also serves as chair of
	board zero otherwise

Appendix B. Detailed procedure to calculate the proxy for real earnings management following Roychowdhury (2006)

Roychowdhury (2006) divides the real activities manipulation into three categories: sales manipulation, reduction of discretionary expenditures, and reduction of Cost of Goods Sold (COGS) due to overproduction and measures real earnings management as abnormal level of operational cash flow, production cost, and discretionary expenditures.

The normal operational cash flow is modeled as:

$$\frac{CFO_t}{At_{t-1}} = \alpha + \beta_1 \times \frac{1}{At_{t-1}} + \beta_2 \times \frac{S_t}{At_{t-1}} + \beta_3 \times \frac{\Delta S_t}{\Delta At_{t-1}} + \mathcal{E}_t$$
(8)

where CFO_t = operating cash flow for a firm at year t; At_{t-1} = a firm's total assets at year t-1; S_t = net sales of a firm at year t; ΔS_t = change in net sale of a firm at year t.

The normal production cost is modeled as:

$$\frac{PROD_t}{At_{t-1}} = \alpha + \beta_1 \times \frac{1}{At_{t-1}} + \beta_2 \times \frac{S_t}{At_{t-1}} + \beta_3 \times \frac{\Delta S_t}{At_{t-1}} + \beta_4 \times \frac{\Delta S_{t-1}}{\Delta At_{t-1}} + \mathcal{E}_t$$
(9)

where PROD_t =cost of goods sold of a firm in year t plus change in inventory of a firm in year t;

The discretionary expenditure is modeled as:

$$\frac{DisExp_t}{At_{t-1}} = \alpha + \beta_1 \times \frac{1}{At_{t-1}} + \beta_2 \times \frac{S_{t-2}}{At_{t-1}} + \mathcal{E}_t$$
(10)

where $DisExp_t = total$ amount of selling, general, and administrative expenses, R&D, advertising and R&D expenses.

The regressions (8), (9), and (10) are run by each year and industry (proxied for by 2digits SIC code). The "abnormal" measures are defined as actual amounts minus the "normal" amounts predicted by the three formulas above.

Following Roychowdhury, (2006) and Cohen et al., (2008), we define real earnings management as abnormal PROD minus abnormal C/F and abnormal DisExp. The higher values indicate a higher level of real earnings management. We only keep the data with more than 15 observations by each year and industry.

Variable	Ν	Mean	SD	P25	P50	P75
Dis Accruals	12935	0.06	0.21	-0.02	0.05	0.14
Real EM	12935	-0.07	0.35	-0.25	-0.06	0.10
Future ROA	12935	-0.01	0.23	-0.03	0.04	0.08
Future DV Increase	12935	0.23	0.42	0.00	0.00	0.00
Stock Return	12935	0.18	0.74	-0.24	0.05	0.37
Degree	12935	64.31	17.34	51.00	63.00	78.00
Eigenvector	12935	57.00	23.24	39.00	58.00	76.00
PCA	12935	0.00	1.32	-0.99	-0.04	1.04
Information Channel	12847	0.07	0.95	-0.63	0.08	0.80
Reputation Channel	12847	0.33	0.46	-0.07	0.41	0.54
Treat_Degree	156	0.23	0.42	0.00	0.00	0.00
Treat_Eigenvector	248	0.32	0.47	0.00	0.00	1.00
Treat_PCA	250	0.26	0.44	0.00	0.00	1.00
Post	696	0.50	0.50	0.00	0.50	1.00
Operating CF	12935	0.06	0.18	0.02	0.08	0.14
NA Accruals	12935	-0.13	0.15	-0.18	-0.11	-0.05
Size	12935	6.01	1.98	4.58	5.98	7.43
BTM	12935	0.56	0.89	0.26	0.46	0.76
ROA	12935	-0.01	0.23	-0.02	0.04	0.08
Leverage	12935	0.18	0.21	0.00	0.13	0.28
ROAstd	12935	0.09	0.19	0.02	0.04	0.09
CFstd	12935	0.07	0.09	0.03	0.05	0.08
Cycle	12935	0.08	0.65	0.04	0.08	0.13
Sales Growth	12935	0.19	1.37	-0.03	0.07	0.19
Big4	12935	0.73	0.44	0.00	1.00	1.00
Duality	12935	0.42	0.49	0.00	0.00	1.00
Tenure	12935	5.11	5.61	1.00	3.00	7.00

Table 1. Summary Statistics.

Table 2. CEO Social Capital and the use of discretionary accruals. This table presents the results of OLS regressions of the effect CEO social capital, on the level of discretionary accruals used in a firm. The dependent variable (Dis Accruals) is defined as the error term estimated by the cross-sectional Modified Jones (1991) Model. CEO social capital is measured by *Degree* in column (1), *Eigenvector* in column (2), and *PCA* in column (3). All regressions include control variables as *Real EM* (real earnings management estimated by Roychowdhury (2006) Model), Size (natural log of total asset), BTM (Book-to-market equity ratio), ROA (return on asset), Leverage (total debt scaled by total asset), ROAstd (rolling standard deviation of return on assets for the past three years including the current year), CFstd (rolling standard deviation of operating cash flow on assets for past three years including current year), Cycle (days of Operating cycle divided by one thousand), Sales Growth (sales growth in year t), Big4 (indicator variable that equals one if audited by big four auditors), Duality (indicator variable that equals one if a CEO also serves as the board chair in year t), and Tenure (numbers of years a CEO is in the position). All regressions include time & industry fixed effects, and the errors are robust to firm heteroscedasticity. T-value is reported in parentheses. Statistical significance of the coefficients is designated as ***, **, and * at 1%, 5%, and 10% levels, respectively.

Sample		Entire Sample	
Dependent Variable		Dis Accruals	
	(1)	(2)	(3)
Degree	0.000423***		
	(2.63)		
Eigenvector		0.000217*	
		(1.93)	
PCA			0.00533**
			(2.37)
Real EM	0.121***	0.121***	0.121***
	(12.77)	(12.72)	(12.73)
Size	-0.0194***	-0.0182***	-0.0192***
	(-6.85)	(-6.89)	(-6.68)
BTM	-0.00263	-0.00283	-0.00267
	(-1.52)	(-1.63)	(-1.55)
ROA	0.563***	0.562***	0.563***
	(11.14)	(11.12)	(11.11)
Leverage	0.0752***	0.0741***	0.0747***
	(5.27)	(5.26)	(5.26)
ROAstd	-0.0563*	-0.0563*	-0.0563*
	(-1.74)	(-1.74)	(-1.74)
CFstd	0.230***	0.230***	0.230***
	(3.75)	(3.75)	(3.75)
Cycle	-0.00217	-0.00211	-0.00214
	(-0.94)	(-0.93)	(-0.93)
Sales Growth	0.00422	0.00420	0.00422
	(1.08)	(1.07)	(1.07)
Big4	0.00255	0.00238	0.00218
	(0.57)	(0.54)	(0.49)
Duality	0.000851	0.00125	0.000958
	(0.26)	(0.39)	(0.30)
Tenure	-0.000445	-0.000488*	-0.000455
	(-1.53)	(-1.65)	(-1.57)
_cons	0.0259	0.0302	0.0502*
	(1.06)	(1.22)	(1.81)
N	12935	12935	12935
adj. R-sq	0.340	0.340	0.340
Industry Fixed Effect included	Yes	Yes	Yes
Year Fixed Effect Included	Yes	Yes	Yes

t statistics in parentheses

* p<.10,**p<.05,***<0.1

Table 3. CEO Social Capital and Real Earnings Management. This table presents the results of OLS regressions of the effect of CEO social capital, on the level of real earnings management used in a firm. The dependent variable (*Real EM*) is defined by the sum of abnormal prediction cost, abnormal cash flow and abnormal discretionary expense, suggested by Roychowdhury (2006) Model. CEO social capital is measured by *Degree* in column (1), *Eigenvector* in column (2), and *PCA* in column (3). All regressions include control variables as Dis Accruals (the level of discretionary accruals used in a firm estimated by the crosssectional Modified Jones (1991) Model), Size (natural log of total asset), BTM (Book-tomarket equity ratio), ROA (return on asset), Leverage (total debt scaled by total asset), ROAstd (rolling standard deviation of return on assets for the past three years including the current year), CFstd (rolling standard deviation of operating cash flow on assets for past three years including current year), Cycle (days of Operating cycle divided by one thousand), Sales Growth (sales growth in year t), Big4 (indicator variable that equals one if audited by big four auditors), *Duality* (indicator variable that equals one if a CEO also serves as the board chair in year t), and Tenure (numbers of years a CEO is in the position). All regressions include time & industry fixed effects, and the errors are robust to firm heteroscedasticity. T-value is reported in parentheses. Statistical significance of the coefficients is designated as ***, **, and * at 1%, 5%, and 10% levels, respectively.

Sample		Entire Sample	
Dependent Variable		Real EM	
	(1)	(2)	(3)
Degree	-0.000905***		
	(-4.06)		
Eigenvector		-0.000728***	
		(-4.82)	
PCA			-0.0144***
			(-4.94)
Dis Accruals	0.398***	0.397***	0.398***
	(12.18)	(12.18)	(12.19)
Size	0.0184***	0.0175***	0.0192***
	(7.08)	(7.32)	(7.47)
BTM	0.0407***	0.0408***	0.0405***
	(9.38)	(9.40)	(9.36)
ROA	-0.763***	-0.763***	-0.765***
	(-22.68)	(-22.72)	(-22.68)
Leverage	0.0733***	0.0756***	0.0740***
	(4.23)	(4.37)	(4.27)
ROAstd	-0.115***	-0.115***	-0.115***
	(-2.83)	(-2.83)	(-2.82)
CFstd	0.120*	0.120*	0.120*
	(1.66)	(1.67)	(1.67)
Cycle	-0.00196	-0.00210	-0.00202
	(-0.72)	(-0.79)	(-0.75)
Sales Growth	0.00229	0.00229	0.00226
	(0.43)	(0.43)	(0.42)
Big4	-0.0584***	-0.0566***	-0.0568***
	(-7.58)	(-7.34)	(-7.37)
Duality	0.00368	0.00323	0.00381
	(0.64)	(0.57)	(0.67)
Tenure	-0.00186***	-0.00182***	-0.00188***
	(-3.65)	(-3.56)	(-3.69)
_cons	-0.0140	-0.0182	-0.0750**
	(-0.38)	(-0.49)	(-1.99)
N	12935	12935	12935
adj. R-sq	0.245	0.245	0.245
Industry Fixed Effect included	Yes	Yes	Yes
Year Fixed Effect Included	Yes	Yes	Yes

t statistics in parentheses

* p<.10,**p<.05,***<0.1

Table 4. Income increasing (positive) vs. Income decreasing (negative) discretionary accruals. This table presents the results of OLS regressions of the effect of CEO social capital, on the level of income increasing (positive) and income decreasing (negative) discretionary accruals used in a firm. The dependent variable (Dis Accruals) is defined as the error term estimated by the cross-sectional Modified Jones (1991) Model. CEO social capital is measured by Degree in columns (1) and (2), Eigenvector in columns (3) and (4), and PCA in columns (5) and (6). All regressions include control variables as Real EM (real earnings management estimated by Roychowdhury (2006) Model), Size (natural log of total asset), BTM (Book-to-market equity ratio), ROA (return on asset), Leverage (total debt scaled by total asset), *ROAstd* (rolling standard deviation of return on assets for the past three years including the current year), CFstd (rolling standard deviation of operating cash flow on assets for past three years including current year), Cycle (days of Operating cycle divided by one thousand), Sales Growth (sales growth in year t), Big4 (indicator variable that equals one if audited by big four auditors), *Duality* (indicator variable that equals one if a CEO also serves as the board chair in year t), and Tenure (numbers of years a CEO is in the position). All regressions include time & industry fixed effects, and the errors are robust to firm heteroscedasticity. T-value is reported in parentheses. Statistical significance of the coefficients is designated as ***, **, and * at 1%, 5%, and 10% levels, respectively.

Dependent VariableIDB ACTURE(1)(2)(3)(4)(5)(6)Degree0.00001400.000571***10.0003100.000549**10.0114***Equivación(10.1)(2.8)0.0003100.000549**0.00154**Equivación10.0101***0.0003100.000549**0.00154**PCA110.0003100.00049**0.00154**0.0114***PCA110.0514**0.0114***0.0114***Ral FM0.0539***0.0124***0.0114***0.0024***0.0024***CA0.0114***0.0003**0.0134***0.0104***0.0024***CA0.0124***0.0124***0.0024***0.0014**0.0024***CA0.0124***0.0204***0.0005**0.0005***0.0005***CA0.0124***0.0005***0.0005***0.0005***0.0005***CA0.0124***0.0005***0.0005***0.0005***0.0005***CA0.0124***0.0005***0.0005***0.0005***0.0005***CA0.0124***0.0214***0.0214***0.0214***0.0214***CA0.0147***0.0147***0.0147***0.0147***0.0147***CA0.0147***0.0141***0.0141****0.0141**********************************	Sample	Dis Accruals>0 Dis Accruals<0 Dis Accruals>0 Dis Accruals<0 Dis Accruals>0 Dis Accruals								
(1)(2)(3)(4)(5)(6)Degoe0.0001490.00087***Dagoe0.13)0.280Egenvector.0.0003910.000459**0.0013**PCA.0.00539**0.0054**0.010***0.0014**0.0054**0.0113** </th <th>Dependent Variable</th> <th colspan="8">Dis Accruals</th>	Dependent Variable	Dis Accruals								
Degree 0.000149 0.000877*** (0.13) (2.38) Eigenvector (0.43) (0.49) (2.09) PCA (0.30) 0.0005391 0.000459** (0.49) (2.09) PCA (0.30) (2.53) Real EM 0.0539*** 0.00208*** 0.0118*** 0.0119*** (7.68) (5.74) (7.68) (5.72) (7.67) (5.74) Size -0.0118*** -0.0208*** -0.0120*** -0.0120*** -0.0204*** (9.12) (-3.26) (-9.90) (-3.04) (-9.19) (-3.13) BTM -0.00603*** 0.00838 -0.0662*** 0.00735 -0.0661*** 0.00935 ROA 0.212*** 0.524*** 0.524*** 0.0374*** 0.0973*** 0.0549** (7.81) (6.64) (7.82) (6.62) (7.79) (6.62) Lzverage 0.0273*** 0.011*** 0.0273*** 0.0973*** 0.0973*** 0.0973**** 0.0973**** </th <th></th> <th>(1)</th> <th>(2)</th> <th>(3)</th> <th>(4)</th> <th>(5)</th> <th>(6)</th>		(1)	(2)	(3)	(4)	(5)	(6)			
Bigmentor0.0.130.2.83Eigenvector0.00003910.000459**PCA0.4090.2.09PCA0.539***0.059***0.0314***Real-PM0.539***0.019***0.0540***0.010***0.7.680.7.090.0514***0.0114**0.0214***7.7.800.0118**0.0208***0.010***0.010***0.0111***0.0208***0.010***0.010***0.010***0.9.0003**0.00038-0.00602***0.010***0.010***0.9.0003***0.00038-0.00602***0.010***0.010***0.9.0003***0.03116.3.800.023***0.010***0.030***0.00041**0.0212***0.0212***0.2212***0.2212***0.221***0.0300.3116.3.800.0230.031**0.031***0.031***0.0371***0.0371***0.0212***0.0371***0.0371***0.0371***0.0371***0.0371***0.0371***0.0471**0.0471**0.06710.0371***0.04610.0271**0.0471**0.06710.0671*0.04410.0471*0.0471**0.0471**0.0671*0.0671*0.0510.017**0.051**0.0671*0.0671*0.0671*0.0510.017**0.0071**0.0071*0.0071**0.0071**0.0510.017**0.0071**0.0071**0.0072**0.0072**0.0510.0072**0.0071**0.0072**0.0072**0.0072**0.01	Degree	0.0000149	0.000877***							
Eigenector0.0003910.00459**PCA0.0539**0.0113**PCA0.0539**0.0540**0.0110**Real-EM0.0539**0.019**0.0540**0.0110**7.080.0120**0.0110**0.0540**0.0110**7.080.0111**0.0204**0.0120**0.0110**6.0110.0204**0.0120**0.0120**0.0120**6.0110.00033*0.00003*0.00003*0.00013*0.00013*7.3090.311(3.08)0.0230.00014**0.00014**6.0210.526**0.212**0.524**0.212**0.526**7.3090.311(3.08)0.0373**0.0374**0.093**80A0.0137**0.0373**0.0978**0.0374**0.093***60A310.016*0.0373**0.06920.0374**0.069280A40.0373**0.0478*0.06920.0374**0.069360A340.016*0.0373**0.06920.0374**0.067460A340.016*0.017*0.6210.6310.067460A340.016*0.017*0.6210.017*0.62161A310.016*0.017*0.016*0.00740.0075*62A40.027*0.016*0.007*0.007*0.007*62A40.028*0.00140.008*0.007*0.001462A50.00140.00140.0015*0.0014*0.0015*62A50.0014*0.0016*0.0016*0.00		(0.13)	(2.88)							
PCA0.0900.0113**Rel PM0.053***0.10***0.054***0.010***0.034***0.10**Rel PM0.053***0.10***0.054***0.012***0.018***0.012***0.018***0.012***0.018***0.012***0.018***0.012***0.0118***0.0018***0.0118***0.0118***0.0118***0.0118***0.0118***0.0118***0.0118***0.0118***0.0118***0.0118***0.0118***0.0118***0.0118***0.0118***0.0118***0.0118**0.0118***0.0118***0.0118***0.0118***0.0118***0.0118***0.0118***0.0118***0.0118***0.0018****0.0018****0.0018****0.0018****0.0018***** <td>Eigenvector</td> <td></td> <td></td> <td>0.0000391</td> <td>0.000459**</td> <td></td> <td></td>	Eigenvector			0.0000391	0.000459**					
PCA0.00345*0.101***0.0539***0.109***0.0540***0.110***0.0540***0.110***Real EM0.0539***0.109***0.0540***0.101***0.0540***0.0118***0.0018***0.0118***0.0018***0.0118***0.0018***0.0018***0.0018***0.00084**0.00084**0.00084**0.00084**0.00084**0.00084**0.00084**0.00094***0.00094***0.021***0.057***0.067**0.057***0.067***0.057*				(0.49)	(2.09)					
Real EM 0.0539^{***} 0.109^{***} 0.0540^{***} 0.110^{***} 0.0540^{***} 0.110^{***} Real EM 0.0539^{***} 0.0208^{***} 0.0120^{***} 0.0120^{***} 0.0120^{***} 0.0204^{***} Size -0.0118^{***} -0.0208^{***} 0.0007^{***} 0.0185^{***} -0.0120^{***} 0.00091^{***} BTM -0.00603^{***} 0.00003^{***} 0.000735 0.00001^{***} 0.00091^{***} 0.00091^{***} ROA 0.212^{***} 0.526^{***} 0.212^{***} 0.524^{***} 0.212^{***} 0.526^{***} ROA 0.212^{***} 0.526^{***} 0.0073^{***} 0.0373^{***} 0.0978^{***} 0.0374^{***} ROA 0.212^{***} 0.526^{***} 0.027^{***} 0.0373^{***} 0.0978^{***} 0.0374^{***} ROA 0.212^{***} 0.526^{***} 0.077^{**} 0.0374^{***} 0.0978^{**} 0.0374^{***} 0.0993^{***} ROA 0.212^{***} 0.526^{***} 0.0373^{***} 0.0778^{**} 0.0374^{***} 0.0993^{***} ROA 0.212^{***} 0.101^{***} 0.0373^{***} 0.0374^{***} 0.0374^{***} 0.0993^{***} ROA 0.0478^{**} 0.0661 0.238^{***} 0.0073 0.00871^{**} 0.00871^{**} ROAsid 0.0078^{**} 0.0061^{**} 0.0088^{**} 0.00871^{**} 0.00871^{**} 0.00871^{**} CFad 0.00808^{**} 0.0081^{**} 0.0081^{**} 0.0081^{**} 0.0081^{**} <	PCA					0.000545	0.0113**			
Real EM 0.0539*** 0.109*** 0.0540*** 0.110*** 0.0540*** 0.110*** Size -0.0118*** -0.0208*** -0.0120*** -0.0185*** -0.0120*** -0.0120*** Size -0.0118*** -0.0208*** -0.0162*** -0.0185*** -0.0120*** -0.0120*** Size -0.01603*** 0.00603*** 0.00602*** 0.000735 -0.00601*** 0.000945 Size -0.0212*** 0.526*** 0.212*** 0.524*** 0.212*** 0.526*** ROA 0.212*** 0.526*** 0.212*** 0.0373*** 0.00978** 0.0093*** ROA 0.212*** 0.526*** 0.0373*** 0.0978*** 0.0374*** 0.0993*** Leverage 0.0373*** 0.101*** 0.0373*** 0.0978** 0.0078* -0.0987 ROA 0.217 (-1.61) (2.17) (-1.62) (2.17) (-1.62) CFsid 0.236*** 0.0661 0.236*** 0.0692 0.00826 0.00871 Cyck						(0.36)	(2.53)			
Size (7.68) (5.74) (7.68) (5.72) (7.67) (5.74) Size -0.0118^{***} -0.0208^{***} -0.0120^{***} -0.0125^{***} -0.0124^{***} -0.0204^{***} ITM -0.00603^{***} 0.000338 -0.00602^{***} 0.000735 -0.00601^{***} 0.000945^{**} ROA 0.212^{***} 0.526^{***} 0.212^{***} 0.524^{***} 0.212^{***} 0.526^{***} ROA 0.212^{***} 0.526^{***} 0.212^{***} 0.524^{***} 0.212^{***} 0.526^{***} ROA 0.212^{***} 0.526^{***} 0.0373^{***} 0.0373^{***} 0.0978^{***} 0.0374^{***} 0.0993^{***} ROA 0.0373^{***} 0.101^{***} 0.0373^{***} 0.0978^{***} 0.0374^{***} 0.0993^{***} ROAsd 0.0478^{**} -0.0977 0.0478^{**} -0.0973 0.0478^{**} -0.0987 ROAsd 0.0478^{**} -0.0977 0.0478^{**} 0.0923^{***} 0.0673 CFsd 0.236^{***} 0.0661 0.236^{***} 0.0673 0.0072 Cfsd 0.00828 0.0941 0.00830 0.0072 0.00826 0.00871 Cyck 0.008084^{***} -0.00602 0.0084^{***} -0.00602 0.0084^{***} -0.00628 Cyck 0.008084^{***} -0.00602 0.0084^{***} -0.00628 0.017^{**} -0.00612 0.0084^{***} -0.00621 Dashy 0.008084^{***} -0.00669 0.017^{**} $(-1.$	Real EM	0.0539***	0.109***	0.0540***	0.110***	0.0540***	0.110***			
Size -0.0118*** -0.0208*** -0.0120*** -0.0120*** -0.0120*** -0.0120*** ISTM -0.00603*** 0.000838 -0.00602*** 0.000735 -0.00601*** 0.00093 BTM -0.00603*** 0.0013 (-3.08) (0.28) (-3.07) (0.36) ROA 0.212*** 0.526*** 0.212*** 0.524*** 0.212*** 0.526*** (7.81) (6.64) (7.82) (6.62) (7.79) (6.62) Leverage 0.0373*** 0.011*** 0.0373*** 0.0978*** 0.0374*** 0.0993*** ROAstd 0.0478** -0.0977 0.0478** -0.0987 (4.07) (1.61) ROAstd 0.0478** -0.0977 0.0478** -0.0987 0.0678 0.0678 CStd 0.236*** 0.0661 0.236*** 0.0692 0.236*** 0.0087 CStd 0.236*** 0.0661 0.236*** 0.0692 0.00826 0.0087 Cyck 0.00804*** -0.00602 0.008084**		(7.68)	(5.74)	(7.68)	(5.72)	(7.67)	(5.74)			
(-9.12) (-3.26) (-9.00) (-3.04) (-9.19) (-3.13) BTM -0.06603^{***} 0.000838 -0.0602^{***} 0.00735 -0.00601^{***} 0.00945 ROA 0.212^{***} 0.526^{***} 0.212^{***} 0.524^{***} 0.212^{***} 0.526^{***} (-7.81) (6.64) (7.82) (6.62) (7.79) (6.62) Leverage 0.0373^{***} 0.01^{***} 0.0373^{***} 0.0374^{***} 0.0993^{***} (-3.81) (4.08) (3.82) (4.07) (3.81) (4.07) ROAstd 0.0478^{**} -0.0977 0.0478^{**} -0.0993 0.0478^{**} -0.0987 ROAstd 0.0478^{**} -0.0977 0.0478^{**} -0.0993 0.0478^{**} -0.0987 CFstd 0.236^{***} 0.0661 0.236^{***} 0.0692 0.236^{***} 0.0673 Cycle 0.00828 0.00941 0.00830 0.0792 0.00826 0.00871 Cycle 0.00828 0.00941 0.00830 0.0792 0.00826 0.00871 Cycle 0.00804^{***} -0.00692 0.0084^{***} -0.00692 0.00826 0.00871 Cycle 0.00804^{***} -0.00704^{**} 0.0167^{**} -0.00698^{**} 0.0162^{**} Cycle 0.00837 -0.0072^{***} -0.0072^{***} -0.0072^{***} -0.0072^{***} Dualiy 0.00387 -0.0071^{*} -0.0072^{**} -0.0072^{***} -0.00072^{***} -0.0072^{*	Size	-0.0118***	-0.0208***	-0.0120***	-0.0185***	-0.0120***	-0.0204***			
BTM -0.00603*** 0.000838 -0.00602*** 0.000735 -0.00601*** 0.000945 ROA 0.212*** 0.526*** 0.212*** 0.524*** 0.212*** 0.526*** ROA 0.0373*** 0.101*** 0.0373*** 0.00735*** 0.0373*** 0.00978*** 0.0374*** 0.0993*** Leverage 0.0373*** 0.101*** 0.0373*** 0.0078*** 0.0374*** 0.0993*** ROAstd 0.0478** -0.0977 0.0478** -0.0993 0.0478** -0.0987 C519 (1.61) (2.17) (-1.62) (2.17) (-1.62) C519 (0.65) (5.19) (0.68) (5.19) (0.66) Cycle 0.00828 0.00941 0.00830 0.00792 0.00826 0.00871 Sales Growth 0.00804*** -0.00602 0.0084*** -0.00612 0.0084** -0.00692 0.0162** Ig4 -0.00689* 0.170** -0.00714 0.0167** -0.00698* 0.0162** Ig4		(-9.12)	(-3.26)	(-9.90)	(-3.04)	(-9.19)	(-3.13)			
(-3.09) (0.31) (-3.08) (0.28) (-3.07) (0.36) ROA 0.212^{***} 0.524^{***} 0.524^{***} 0.212^{***} 0.524^{***} 0.524^{***} 0.524^{***} 0.524^{***} 0.524^{***} 0.524^{***} 0.524^{***} 0.524^{***} 0.524^{***} 0.524^{***} 0.524^{***} 0.524^{***} 0.524^{***} 0.0973^{***} 0.0973^{***} 0.0973^{***} 0.0973^{***} 0.0973^{***} 0.0973^{***} 0.0973^{***} 0.0973^{***} 0.0973^{***} 0.0973^{***} 0.0973^{***} 0.0973^{***} 0.0973^{***} 0.0973^{***} 0.0973^{***} 0.0973^{***} 0.0987^{***} 0.0987^{***} 0.0987^{***} 0.0987^{***} 0.0987^{***} 0.0987^{***} 0.0987^{***} 0.0987^{***} 0.0987^{***} 0.0987^{***} 0.0067^{***} 0.0087^{***} 0.0067^{***} 0.0087^{***} 0.0067^{***} 0.0089^{***} 0.0067^{***} 0.0069^{***} 0.0067^{***} 0.0069^{***} 0.0067^{***} 0.0069^{***} 0.0067^{***} 0.0067^{***} 0.0067^{***} 0.0067^{***} 0.0067^{***} 0.0067^{***} 0.0067^{***} 0.0067^{***} 0.0067^{***} 0.0067^{**	BTM	-0.00603***	0.000838	-0.00602***	0.000735	-0.00601***	0.000945			
ROA 0.212^{***} 0.526^{***} 0.212^{***} 0.524^{***} 0.212^{***} 0.526^{***} Leverage 0.0373^{***} 0.101^{***} 0.0373^{***} 0.0978^{***} 0.0374^{***} 0.0993^{***} ROAsd 0.0478^{**} -0.0977 0.0478^{**} -0.0993 0.0478^{**} -0.0987 ROAsd 0.0478^{**} -0.0977 0.0478^{**} -0.0993 0.0478^{**} -0.0987 ROAsd 0.0478^{**} -0.0977 0.0478^{**} -0.0993 0.0478^{**} -0.0987 ROAsd 0.0478^{**} 0.0093 0.0478^{**} -0.0987 0.0478^{**} 0.0993 0.0478^{**} -0.0987 CStd 0.236^{***} 0.0661 0.236^{***} 0.0673 0.0692 0.236^{***} 0.0673 Cycle 0.0008028 0.00941 0.000830 0.00720^{**} 0.000826 0.0081^{***} Sales Growth 0.000804^{***} -0.00612 0.00804^{***} 0.000612 0.00804^{***} <t< td=""><td></td><td>(-3.09)</td><td>(0.31)</td><td>(-3.08)</td><td>(0.28)</td><td>(-3.07)</td><td>(0.36)</td></t<>		(-3.09)	(0.31)	(-3.08)	(0.28)	(-3.07)	(0.36)			
(7.81) (6.64) (7.82) (6.62) (7.79) (6.62) Leverage 0.0373^{***} 0.101^{***} 0.0373^{***} 0.0978^{***} 0.0374^{***} 0.0993^{***} ROAsid 0.0478^{**} -0.0977 0.0478^{**} -0.0993 0.0478^{**} -0.0987 ROAsid 0.0478^{**} -0.0977 0.0478^{**} -0.0993 0.0478^{**} -0.0987 (2.17) (-1.61) (2.17) (-1.62) (2.17) (-1.62) CFstd 0.236^{***} 0.0661 0.236^{***} 0.0692 0.236^{***} 0.0673 Cycle 0.00828 0.00941 0.00830 0.00792 0.00826 0.00871 Cycle 0.00804^{***} -0.00602 0.00804^{***} -0.00612 0.00804^{***} -0.00602 Sales Growth 0.00804^{***} -0.00602 0.00804^{***} -0.00612 0.00804^{***} -0.00602 Sales Growth 0.00804^{***} -0.00602 0.00804^{***} -0.00612 0.00804^{***} -0.00612 Sales Growth 0.00804^{***} -0.00602 0.00704^{**} -0.00673 0.00673 0.00673 Dualiy 0.00837 -0.00738 0.00383 -0.00728^{**} 0.00720^{**} 0.000221 Dualiy 0.00387 -0.00714 -0.00720^{**} -0.000720^{**} 0.00021 Tenue -0.000725^{**} -0.000714 -0.00728^{**} -0.000720^{**} 0.00021 _cords (-1.38) (1.29) $(-1.$	ROA	0.212***	0.526***	0.212***	0.524***	0.212***	0.526***			
Leverage 0.0373*** 0.101*** 0.0373*** 0.0978*** 0.0374*** 0.0993*** ROAsid 0.0478** -0.0977 0.0478** -0.0993 0.0478** -0.0987 ROAsid 0.0478** -0.0977 0.0478** -0.0993 0.0478** -0.0987 CFsid 0.236*** 0.0661 0.236*** 0.0692 0.236*** 0.0673 CFsid 0.236*** 0.0692 0.236*** 0.0673 0.0673 CS19 (0.65) (5.19) 0.688 (5.19) (0.66) Cycle 0.000828 0.00941 0.000830 0.00722 0.000826 0.0081 Sales Growth 0.00804*** -0.00602 0.00804*** -0.00612 0.00804*** -0.00602 Big4 -0.00689* 0.0170** -0.00714 0.0167* -0.00698* 0.0162** Leiniy (1.30) (1.38) (1.29) (1.20) (1.34) (2.17) Dualiy 0.00387 -0.00738 0.00383 -0.00678		(7.81)	(6.64)	(7.82)	(6.62)	(7.79)	(6.62)			
(3.81) (4.08) (3.82) (4.07) (3.81) (4.07) ROAsid 0.0478** -0.0977 0.0478** -0.0993 0.0478** -0.0987 (2.17) (-1.61) (2.17) (-1.62) (2.17) (-1.62) CFstd 0.236*** 0.0661 0.236*** 0.0692 0.236*** 0.0673 (5.19) (0.65) (5.19) (0.68) (5.19) (0.66) Cycle 0.00828 0.00941 0.00830 0.00720 0.00804 0.00621 Sales Growth 0.00804*** -0.00602 0.00804*** -0.00612 0.00804*** -0.00602 Sales Growth 0.00804*** -0.00602 0.00804*** -0.00612 0.00804*** -0.00602 Big4 -0.00689* 0.0170** -0.00714* 0.0167** -0.00698* 0.0162** Duality 0.00387 -0.00738 0.00383 -0.00678 0.00326 -0.000720*** -0.00021 _cons (-1.30) (-1.38) (1.29) (-1.3	Leverage	0.0373***	0.101***	0.0373***	0.0978***	0.0374***	0.0993***			
ROAsid 0.0478** -0.0977 0.0478** -0.0993 0.0478** -0.0987 (2.17) (-1.61) (2.17) (-1.62) (2.17) (-1.62) CFsid 0.236*** 0.0661 0.236*** 0.0692 0.236*** 0.0673 (5.19) (0.65) (5.19) (0.68) (5.19) (0.66) Cycle 0.008028 0.00941 0.008030 0.00792 0.00826 0.00871 (1.01) (0.32) (1.01) (0.27) (1.00) (0.30) Sales Growth 0.00804*** -0.00602 0.0084*** -0.00612 0.00804*** -0.0062 (3.04) (-0.74) (3.04) (-0.75) (3.04) (-0.74) Big4 -0.00689* 0.0170** -0.00678 0.00383 -0.00720 Duality 0.00387 -0.00738 0.00383 -0.00678 0.000326 -0.00720*** -0.000221 _cons (1.30) (-1.38) (1.29) (-1.65) (5.45) 0.0152		(3.81)	(4.08)	(3.82)	(4.07)	(3.81)	(4.07)			
(2.17) (-1.61) (2.17) (-1.62) (2.17) (-1.62) CFstd 0.236^{***} 0.0661 0.236^{***} 0.0692 0.236^{***} 0.0673 (5.19) (0.65) (5.19) (0.68) (5.19) (0.66) Cycle 0.000828 0.00941 0.000830 0.00792 0.00826 0.00871 Cycle (1.01) (0.32) (1.01) (0.27) (1.00) (0.30) Sales Growth 0.00804^{***} -0.00602 0.00804^{***} -0.00612 0.00804^{***} -0.00602 Big4 $-0.00689*$ 0.0170^{**} -0.00704^{*} 0.0167^{**} -0.00698^{*} 0.0162^{**} Duality 0.00387 -0.00738 0.00383 -0.00720^{**} -0.000720^{**} -0.000720^{**} -0.000221 Tenure -0.00725^{***} -0.000714^{**} -0.00326 -0.00720^{***} -0.00021 _cons 0.117^{***} -0.0350 0.117^{***} -0.0278 0.119^{***} 0.0152 _cons 0.117^{***} -0.0350 0.117^{***} -0.0278 0.119^{***} 0.0152 N 8833 4102 8833 4102 8833 4102 adj. R-sq 0.219 0.469 0.219 0.459 0.219 0.460 Industry Fixed Effect includedYesYesYesYesYesYes	ROAstd	0.0478**	-0.0977	0.0478**	-0.0993	0.0478**	-0.0987			
CFstd 0.236^{***} 0.0661 0.236^{***} 0.0692 0.236^{***} 0.0673 Cycle 0.000828 0.00941 0.000830 0.00792 0.000826 0.00871 Cycle 0.000828 0.00941 0.000830 0.00792 0.000826 0.00871 (1.01) (0.32) (1.01) (0.27) (1.00) (0.30) Sales Growth 0.00804^{***} -0.00602 0.00804^{***} -0.00612 0.00804^{***} -0.00602 Big4 -0.00689^{*} 0.0170^{**} -0.00704^{*} 0.0167^{**} -0.00698^{*} 0.0162^{**} Duality 0.00387 -0.00738 0.00383 -0.00728 -0.00722 Curse -0.000725^{***} -0.00174 -0.00720^{***} -0.000720^{***} -0.000221 _cons 0.117^{***} -0.0350 0.117^{***} -0.0278 0.119^{***} 0.0152 N8833 4102 8833 4102 8833 4102 8833 4102 Industry Fixed Effect includedYesYesYesYesYesYesYes		(2.17)	(-1.61)	(2.17)	(-1.62)	(2.17)	(-1.62)			
(5.19) (0.65) (5.19) (0.68) (5.19) (0.66) Cycle 0.000828 0.00941 0.000830 0.00792 0.000826 0.00871 (1.01) (0.32) (1.01) (0.27) (1.00) (0.30) Sales Growth 0.00804^{***} -0.00602 0.00804^{***} -0.00612 0.00804^{***} (3.04) (-0.74) (3.04) (-0.75) (3.04) (-0.74) Big4 -0.00689^{**} 0.0170^{**} -0.00704^{**} 0.0167^{**} -0.00698^{**} 0.0162^{**} (-1.82) (2.27) (-1.86) (2.24) (-1.84) (2.17) Duality 0.00387 -0.00738 0.00383 -0.00728 0.00722 (1.30) (-1.38) (1.29) (-1.26) (1.29) (-1.34) Tenure -0.000725^{***} -0.000174 -0.00720^{***} -0.000720^{***} -0.000221 (-2.98) (-0.36) (-2.97) (-0.65) (-2.96) (-0.46) $_cons$ 0.117^{***} -0.0350 0.117^{***} -0.0278 0.119^{***} 0.0152 N 8833 4102 8833 4102 8833 4102 $adj. R-sq$ 0.219 0.460 0.219 0.459 0.219 0.460 Industry Fixed Effect includedYesYesYesYesYesYesYes	CFstd	0.236***	0.0661	0.236***	0.0692	0.236***	0.0673			
Cycle 0.000828 0.00941 0.000830 0.00792 0.000826 0.00871 (1.01) (0.32) (1.01) (0.27) (1.00) (0.30) Sales Growth $0.00804***$ -0.00602 $0.00804***$ -0.00612 $0.00804***$ -0.00602 (3.04) (-0.74) (3.04) (-0.75) (3.04) (-0.74) Big4 $-0.00689*$ $0.0170**$ $-0.00704*$ $0.0167**$ $-0.00698*$ $0.0162**$ (-1.82) (2.27) (-1.86) (2.24) (-1.84) (2.17) Duality 0.00387 -0.00738 0.00383 -0.00678 0.00383 -0.00722 Tenure $-0.000725***$ -0.00174 $-0.00720***$ $-0.000720***$ $-0.000720***$ -0.000221 (-2.98) (-0.36) (-2.97) (-0.65) (-2.96) (-0.46) $_cons$ $0.117***$ -0.0350 $0.117***$ -0.0278 $0.119***$ 0.0152 $_adj. R-sq$ 0.219 0.460 0.219 0.459 0.219 0.460 Industry Fixed Effect includedYesYesYesYesYesYesYes		(5.19)	(0.65)	(5.19)	(0.68)	(5.19)	(0.66)			
(1.01) (0.32) (1.01) (0.27) (1.00) (0.30) Sales Growth 0.00804^{***} -0.00602 0.00804^{***} -0.00612 0.00804^{***} -0.00602 (3.04) (-0.74) (3.04) (-0.75) (3.04) (-0.74) Big4 -0.00689^* 0.0170^{**} -0.00704^* 0.0167^{**} -0.00698^* 0.0162^{**} (-1.82) (2.27) (-1.86) (2.24) (-1.84) (2.17) Duality 0.00387 -0.00738 0.00383 -0.00678 0.00383 -0.00722 (1.30) (-1.38) (1.29) (-1.26) (1.29) (-1.34) Tenure -0.000725^{***} -0.000174 -0.000720^{***} -0.000221 (-2.98) (-0.36) (-2.97) (-0.65) (-2.96) (-0.46) $_cons$ 0.117^{***} -0.0350 0.117^{***} -0.0278 0.119^{***} 0.0152 (5.41) (-0.82) (5.41) (-0.65) (5.35) (0.29) N 8833 4102 8833 4102 8833 4102 adj. R-sq 0.219 0.460 0.219 0.459 0.219 0.460 Industry Fixed Effect includedYesYesYesYesYesYesYesYes	Cycle	0.000828	0.00941	0.000830	0.00792	0.000826	0.00871			
Sales Growth 0.00804^{***} -0.00602 0.00804^{***} -0.00612 0.00804^{***} -0.00602 Big4 -0.00689^{*} 0.0170^{**} -0.00704^{*} 0.0167^{**} -0.00698^{*} 0.0162^{**} (-1.82) (2.27) (-1.86) (2.24) (-1.84) (2.17) Duality 0.00387 -0.00738 0.00383 -0.00678 0.00383 -0.00722 Tenure -0.000725^{***} -0.000174 -0.00720^{***} -0.000720^{***} -0.000221 (-2.98) (-0.36) (-2.97) (-0.65) (-2.96) (-0.46) $_cons$ 0.117^{***} -0.0350 0.117^{***} -0.0278 0.119^{***} 0.0152 N 8833 4102 8833 4102 8833 4102 8833 4102 Industry Fixed Effect includedYesYesYesYesYesYesYesYes		(1.01)	(0.32)	(1.01)	(0.27)	(1.00)	(0.30)			
(3.04) (-0.74) (3.04) (-0.75) (3.04) (-0.74) Big4 $-0.00689*$ $0.0170**$ $-0.00704*$ $0.0167**$ $-0.00698*$ $0.0162**$ (-1.82) (2.27) (-1.86) (2.24) (-1.84) (2.17) Duality 0.00387 -0.00738 0.00383 -0.00788 0.00383 -0.00722 (1.30) (-1.38) (1.29) (-1.26) (1.29) (-1.34) Tenure $-0.000725***$ -0.000174 $-0.00720***$ -0.000326 $-0.000720***$ -0.000221 (-2.98) (-0.36) (-2.97) (-0.65) (-2.96) (-0.46) $_cons$ $0.117***$ -0.0350 $0.117***$ -0.0278 $0.119***$ 0.0152 N 8833 4102 8833 4102 8833 4102 adj. R-sq 0.219 0.460 0.219 0.459 0.219 0.460 Industry Fixed Effect includedYesYesYesYesYesYesYesYes	Sales Growth	0.00804***	-0.00602	0.00804***	-0.00612	0.00804***	-0.00602			
Big4 -0.00689^* 0.0170^{**} -0.00704^* 0.0167^{**} -0.00698^* 0.0162^{**} Luality (-1.82) (2.27) (-1.86) (2.24) (-1.84) (2.17) Duality 0.00387 -0.00738 0.00383 -0.00678 0.00383 -0.00722 (1.30) (-1.38) (1.29) (-1.26) (1.29) (-1.34) Tenure -0.000725^{***} -0.000174 -0.00720^{***} -0.000720^{***} -0.000221 (-2.98) (-0.36) (-2.97) (-0.65) (-2.96) (-0.46) $_cons$ 0.117^{***} -0.0350 0.117^{***} -0.0278 0.119^{***} 0.0152 (5.41) (-0.82) (5.41) (-0.65) (5.35) (0.29) N 8833 4102 8833 4102 8833 4102 adj. R-sq 0.219 0.460 0.219 0.459 0.219 0.460 Industry Fixed Effect includedYesYesYesYesYesYesYes		(3.04)	(-0.74)	(3.04)	(-0.75)	(3.04)	(-0.74)			
(-1.82) (2.27) (-1.86) (2.24) (-1.84) (2.17) Duality 0.00387 -0.00738 0.00383 -0.00678 0.00383 -0.00722 (1.30) (-1.38) (1.29) (-1.26) (1.29) (-1.34) Tenure -0.000725^{***} -0.000174 -0.000720^{***} -0.000720^{***} -0.000720^{***} _cons (-2.98) (-0.36) (-2.97) (-0.65) (-2.96) (-0.46) _cons 0.117^{***} -0.0350 0.117^{***} -0.0278 0.119^{***} 0.0152 N 8833 4102 8833 4102 8833 4102 adj. R-sq 0.219 0.460 0.219 0.459 0.219 0.460 Industry Fixed Effect includedYesYesYesYesYesYesYesYesYes	Big4	-0.00689*	0.0170**	-0.00704*	0.0167**	-0.00698*	0.0162**			
Duality 0.00387 -0.00738 0.00383 -0.00678 0.00383 -0.00722 (1.30) (-1.38) (1.29) (-1.26) (1.29) (-1.34) Tenure -0.000725*** -0.000174 -0.000720*** -0.000326 -0.000720*** -0.000221 (-2.98) (-0.36) (-2.97) (-0.65) (-2.96) (-0.46) _cons 0.117*** -0.0350 0.117*** -0.0278 0.119*** 0.0152 (5.41) (-0.82) (5.41) (-0.65) (5.35) (0.29) N 8833 4102 8833 4102 8833 4102 adj. R-sq 0.219 0.460 0.219 0.459 0.219 0.460 Industry Fixed Effect included Yes Yes Yes Yes Yes Yes Yes Yes		(-1.82)	(2.27)	(-1.86)	(2.24)	(-1.84)	(2.17)			
(1.30) (-1.38) (1.29) (-1.26) (1.29) (-1.34) Tenure -0.000725^{***} -0.000174 -0.000720^{***} -0.000326 -0.000720^{***} -0.000221 (-2.98) (-0.36) (-2.97) (-0.65) (-2.96) (-0.46) _cons 0.117^{***} -0.0350 0.117^{***} -0.0278 0.119^{***} 0.0152 (5.41) (-0.82) (5.41) (-0.65) (5.35) (0.29) N883341028833410288334102adj. R-sq 0.219 0.460 0.219 0.459 0.219 0.460 Industry Fixed Effect includedYesYesYesYesYesYesYes	Duality	0.00387	-0.00738	0.00383	-0.00678	0.00383	-0.00722			
Tenure -0.000725*** -0.000174 -0.000720*** -0.000326 -0.000720*** -0.000221 _cons (-2.98) (-0.36) (-2.97) (-0.65) (-2.96) (-0.46) _cons 0.117*** -0.0350 0.117*** -0.0278 0.119*** 0.0152 (5.41) (-0.82) (5.41) (-0.65) (5.35) (0.29) N 8833 4102 8833 4102 8833 4102 adj. R-sq 0.219 0.460 0.219 0.459 0.219 0.460 Industry Fixed Effect included Yes Yes Yes Yes Yes Yes		(1.30)	(-1.38)	(1.29)	(-1.26)	(1.29)	(-1.34)			
(-2.98) (-0.36) (-2.97) (-0.65) (-2.96) (-0.46) _cons 0.117*** -0.0350 0.117*** -0.0278 0.119*** 0.0152 (5.41) (-0.82) (5.41) (-0.65) (5.35) (0.29) N 8833 4102 8833 4102 8833 4102 adj. R-sq 0.219 0.460 0.219 0.459 0.219 0.460 Industry Fixed Effect included Yes Yes Yes Yes Yes Yes Yes	Tenure	-0.000725***	-0.000174	-0.000720***	-0.000326	-0.000720***	-0.000221			
cons 0.117*** -0.0350 0.117*** -0.0278 0.119*** 0.0152 (5.41) (-0.82) (5.41) (-0.65) (5.35) (0.29) N 8833 4102 8833 4102 8833 4102 adj. R-sq 0.219 0.460 0.219 0.459 0.219 0.460 Industry Fixed Effect included Yes Yes Yes Yes Yes Yes Yes		(-2.98)	(-0.36)	(-2.97)	(-0.65)	(-2.96)	(-0.46)			
(5.41) (-0.82) (5.41) (-0.65) (5.35) (0.29) N 8833 4102 8833 4102 8833 4102 adj. R-sq 0.219 0.460 0.219 0.459 0.219 0.460 Industry Fixed Effect included Yes Yes Yes Yes Yes Yes	_cons	0.117***	-0.0350	0.117***	-0.0278	0.119***	0.0152			
N 8833 4102 8833 4102 8833 4102 adj. R-sq 0.219 0.460 0.219 0.459 0.219 0.460 Industry Fixed Effect included Yes Yes Yes Yes Yes Yes		(5.41)	(-0.82)	(5.41)	(-0.65)	(5.35)	(0.29)			
adj. R-sq0.2190.4600.2190.4590.2190.460Industry Fixed Effect includedYesYesYesYesYesYes	N	8833	4102	8833	4102	8833	4102			
Industry Fixed Effect included Yes Yes Yes Yes Yes Yes	adj. R-sq	0.219	0.460	0.219	0.459	0.219	0.460			
	Industry Fixed Effect included	Yes	Yes	Yes	Yes	Yes	Yes			
Year Fixed Effect Included Yes Yes Yes Yes Yes Yes	Year Fixed Effect Included	Yes	Yes	Yes	Yes	Yes	Yes			

CEO Social Capital and Discretionary Accruals

t statistics in parentheses

* p<.10,**p<.05,***<0.1

Table 5. Instrumented CEO Social Capital and the use of discretionary accruals. This table presents the second-stage results of the effect of instrumented CEO social capital, on the level of discretionary accruals used in a firm. The dependent variable (*Dis Accruals*) is defined as the error term estimated by the cross-sectional Modified Jones (1991) Model. CEO social capital is measured by *Instrumented Degree* in columns (1) - (3), *Instrumented* Eigenvector in columns (4) - (6), and PCA in columns (7) - (9). All regressions include control variables as *Real EM* (real earnings management estimated by Roychowdhury (2006) Model), Size (natural log of total asset), BTM (Book-to-market equity ratio), ROAstd (rolling standard deviation of return on assets for the past three years including the current year), CFstd (rolling standard deviation of operating cash flow on assets for past three years including current year), Cycle (days of Operating cycle divided by one thousand), Sales Growth (sales growth in year t), Big4 (indicator variable that equals one if audited by big four auditors), Duality (indicator variable that equals one if a CEO also serves as the board chair in year t), and Tenure (numbers of years a CEO is in the position). All regressions include time & industry fixed effects, and the errors are robust to firm heteroscedasticity. T-value is reported in parentheses. Statistical significance of the coefficients is designated as ***, **,

and * at 1%, 5%, and 10% levels, respectively.

Sample	Entire Sample	Dis Accruals > 0	Dis Accruals < 0	Entire Sample	Dis Accruals > 0	Dis Accruals < 0	Entire Sample	Dis Accruals > 0	Dis Accruals < 0
Dependent Variable					Dis Accruals				
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
Instrumented Degree	0.000680**	-0.000324	0.00163***						
	(2.22)	(-1.17)	(3.42)						
Instrumented Eigenvector				0.000151	-0.000121	0.000443*			
				(0.87)	(-0.69)	(1.86)			
PCA							0.00620	-0.00354	0.0161***
							(1.63)	(-0.99)	(2.82)
Real EM	0.122***	0.0532***	0.109***	0.121***	0.0535***	0.110***	0.122***	0.0533***	0.110***
	(12.77)	(7.47)	(5.81)	(12.69)	(7.52)	(5.78)	(12.70)	(7.44)	(5.81)
Size	-0.0209***	-0.00984***	-0.0249***	-0.0179***	-0.0110***	-0.0184***	-0.0195***	-0.0102***	-0.0222***
	(-6.24)	(-5.01)	(-3.69)	(-6.62)	(-7.00)	(-3.23)	(-6.17)	(-5.35)	(-3.48)
BTM	-0.00237	-0.00645***	0.00142	-0.00290*	-0.00616***	0.000715	-0.00260	-0.00633***	0.00128
	(-1.37)	(-3.25)	(0.53)	(-1.67)	(-3.14)	(0.27)	(-1.50)	(-3.19)	(0.49)
ROA	0.566***	0.207***	0.531***	0.561***	0.209***	0.524***	0.564***	0.208***	0.528***
	(11.14)	(7.41)	(6.78)	(11.13)	(7.52)	(6.71)	(11.10)	(7.35)	(6.75)
Leverage	0.0760***	0.0362***	0.103***	0.0741***	0.0371***	0.0978***	0.0748***	0.0367***	0.1000***
	(5.30)	(3.69)	(4.20)	(5.27)	(3.81)	(4.11)	(5.28)	(3.74)	(4.15)
ROAstd	-0.0564*	0.0486**	-0.0972	-0.0562*	0.0480**	-0.0992	-0.0563*	0.0484**	-0.0989*
	(-1.76)	(2.22)	(-1.64)	(-1.74)	(2.19)	(-1.64)	(-1.75)	(2.20)	(-1.65)
CFstd	0.229***	0.236***	0.0628	0.230***	0.236***	0.0692	0.229***	0.236***	0.0662
	(3.76)	(5.21)	(0.63)	(3.76)	(5.22)	(0.68)	(3.76)	(5.22)	(0.66)
Cycle	-0.00220	0.000891	0.0107	-0.00212	0.000831	0.00792	-0.00214	0.000858	0.00907
-	(-0.94)	(1.10)	(0.37)	(-0.93)	(1.02)	(0.27)	(-0.93)	(1.06)	(0.31)
Sales Growth	0.00425	0.00803***	-0.00578	0.00419	0.00804***	-0.00613	0.00423	0.00803***	-0.00590
	(1.09)	(3.06)	(-0.72)	(1.07)	(3.06)	(-0.76)	(1.08)	(3.06)	(-0.73)
Big4	0.00200	-0.00626	0.0150**	0.00271	-0.00630	0.0168**	0.00197	-0.00609	0.0149**
	(0.45)	(-1.64)	(2.03)	(0.60)	(-1.64)	(2.23)	(0.44)	(-1.58)	(1.99)
Duality	0.000401	0.00454	-0.00829	0.00136	0.00413	-0.00676	0.000855	0.00439	-0.00761
	(0.12)	(1.49)	(-1.53)	(0.42)	(1.38)	(-1.27)	(0.26)	(1.45)	(-1.42)
Tenure	-0.000394	-0.000778***	0.0000358	-0.000500*	-0.000749***	-0.000329	-0.000443	-0.000769***	-0.000136
	(-1.36)	(-3.14)	(0.07)	(-1.69)	(-3.08)	(-0.65)	(-1.52)	(-3.13)	(-0.28)
_cons	0.0210	0.124***	-0.0488	0.0313	0.120***	-0.0275	0.0528*	0.107***	0.0297
	(0.85)	(5.48)	(-1.13)	(1.26)	(5.46)	(-0.64)	(1.80)	(4.41)	(0.57)
N	12935	8833	4102	12935	8833	4102	12935	8833	4102
adj. R-sq	0.340	0.218	0.458	0.340	0.219	0.459	0.340	0.219	0.460
Industry Fixed Effect included	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Year Fixed Effect Included	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes

t statistics in parentheses * p<.10,**p<.05,***<0.1

Table 6. Difference-in-difference analysis: Evidence from CEO turnover. This table presents the regression results of difference-in-difference analysis surrounding CEO turnover. We compare the level of discretionary accruals used in firms that switch from CEOs with lower social capital to new CEOs with higher social capital (the treatment group) and firms that switch from CEOs with lower social capital to new CEOs with lower social capital (the control group). The dependent variable (Dis Accruals) is defined as the error term estimated by the cross-sectional Modified Jones (1991) Model. The treatment and control groups are formed based on CEO social capital value measured by Degree in columns (1) - (3), Eigenvector in columns (4) - (6), and PCA in columns (7) - (9). Post is an indicator variable that represents observations following CEO turnover. All regressions include control variables as *Real EM* (real earnings management estimated by Roychowdhury (2006) Model), Size (natural log of total asset), BTM (Book-to-market equity ratio), ROA (return on asset), Leverage (total debt scaled by total asset), ROAstd (rolling standard deviation of return on assets for the past three years including the current year), CFstd (rolling standard deviation of operating cash flow on assets for past three years including current year), Cycle (days of Operating cycle divided by one thousand), *Sales Growth* (sales growth in year t), *Big4* (indicator variable that equals one if audited by big four auditors), *Duality* (indicator variable that equals one if a CEO also serves as the board chair in year t), and Tenure (numbers of years a CEO is in the position). All regressions include time & industry fixed effects, and the errors are robust to firm heteroscedasticity. T-value is reported in parentheses. Statistical significance of the coefficients is designated as ***, **, and * at 1%, 5%, and 10% levels, respectively.

CEO Social Capital and Discretionary Accruals

5	1
J	L

Sample	Reduced Sample for CEO turn	over Dis Accruals > 0	Dis Accruals < 0 Re	duced Sample for CEO turn	over Dis Accruals > 0	Dis Accruals < 0 Re	duced Sample for CEO turn	over Dis Accruals > 0	Dis Accruals < 0
Dependent Variable	•			Ľ	Dis_Accruals		•		
· •	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
Post	0.0207	0.0534	-0.104**	-0.0292	-0.0243	-0.00966	-0.0166	0.0151	-0.110**
	(0.45)	(0.91)	(-3.03)	(-1.01)	(-0.93)	(-0.22)	(-0.56)	(0.47)	(-2.40)
Treat_Degree	0.0181	0.132	-0.287*						
	(0.27)	(1.54)	(-2.65)						
Post × Treat_Degree	-0.0222	-0.0749	0.182**						
	(-0.35)	(-1.07)	(3.93)						
Treat_Eigenvector				0.0158	-0.0240	-0.0156			
				(0.48)	(-0.74)	(-0.27)			
Post × Treat_Eigenvector				0.0349	-0.0233	0.0476			
-				(0.89)	(-0.56)	(0.58)			
Treat_PCA							-0.0249	-0.0403	-0.0897
							(-0.53)	(-0.64)	(-1.52)
Post × Treat_PCA							0.0439	-0.0340	0.150*
							(0.84)	(-0.53)	(1.93)
Real EM	0.0701	0.0920	0.0569	0.160***	0.0368	0.107	0.130**	0.0618	0.0545
	(1.26)	(1.60)	(0.61)	(3.33)	(0.61)	(1.67)	(2.39)	(0.89)	(1.22)
Size	-0.0377***	-0.0679***	-0.0180	-0.0394***	-0.0464***	-0.00654	-0.0267**	-0.0464***	0.00757
	(-2.99)	(-4.85)	(-1.36)	(-4.31)	(-4.29)	(-0.44)	(-2.34)	(-3.64)	(0.39)
BTM	0.0358	-0.0282	0.0527	0.00770	0.00320	0.0321	-0.00317	-0.0625	-0.0275
	(0.75)	(-0.67)	(0.55)	(0.37)	(0.11)	(0.68)	(-0.11)	(-1.30)	(-0.63)
ROA	0.449***	0.434***	0.710***	0.601***	0.404**	0.552***	0.526***	0.380**	0.552***
	(5.33)	(2.96)	(5.93)	(8.88)	(2.39)	(3.82)	(6.76)	(2.48)	(3.84)
Leverage	-0.0408	0.0484	0.161	0.0494	0.202**	0.0395	-0.0902	-0.112	0.0790
C	(-0.45)	(0.27)	(1.29)	(0.57)	(2.27)	(0.31)	(-1.28)	(-0.92)	(0.71)
ROAstd	0.00719	-0.0788	0.724*	-0.0186	0.0446	0.108	-0.0435	-0.0737	0.0670
	(0.06)	(-0.72)	(2.24)	(-0.21)	(0.65)	(0.36)	(-0.44)	(-0.74)	(0.24)
CFstd	-0.0773	0.000217	-0.349	-0.105	-0.159*	-0.267	0.0379	0.0781	-0.494
	(-0.45)	(0.00)	(-0.59)	(-0.69)	(-1.77)	(-0.61)	(0.27)	(0.60)	(-1.30)
Cycle	0.0252	0.289	0.111	0.0192	-0.120	-0.0221	-0.00802	-0.141	0.246
5	(0.13)	(1.10)	(0.24)	(0.13)	(-0.67)	(-0.09)	(-0.04)	(-0.53)	(1.08)
Sales Growth	0.0689	0.0815	0.0296	0.0278	-0.0222	0.0200	0.0771*	0.0679	-0.00145
	(1.41)	(1.37)	(0.38)	(0.74)	(-0.61)	(0.50)	(1.69)	(1.46)	(-0.03)
Big4	0.0842*	0.0797*	0.0738*	-0.000949	-0.00513	-0.0321	0.0258	0.0333	-0.0217
C	(1.90)	(1.78)	(2.13)	(-0.04)	(-0.20)	(-0.64)	(0.83)	(0.99)	(-0.34)
Duality	0.0666	-0.0411	-0.124	0.00313	0.0222	-0.0557	0.0107	0.0353	-0.0396
5	(1.62)	(-0.77)	(-1.36)	(0.13)	(1.09)	(-1.58)	(0.34)	(0.87)	(-0.77)
Tenure	-0.00329	-0.00687	-0.0144**	-0.00291	-0.00193	-0.00184	-0.00620*	-0.00211	-0.00597
	(-0.57)	(-0.60)	(-4.46)	(-1.23)	(-0.44)	(-0.67)	(-1.77)	(-0.23)	(-1.59)
cons	0.311	0.378*	0.422**	0.244**	0.579***	0.0273	0.272**	0.632***	0.0202
-	(1.59)	(1.90)	(2.78)	(2.29)	(5.54)	(0.25)	(2.14)	(4.91)	(0.17)
N	156	103	53	248	160	88	200	133	67
adj. R-sq	0.465	0.346	0.979	0.440	0.373	0.684	0.467	0.302	0.866
Industry Fixed Effect include	ed Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Year Fixed Effect Included	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes

t statistics in parentheses * p<.10,**p<.05,***<0.1

Table 7. CEO Social Capital and the use of discretionary accruals, in the reduced samples predicted by propensity score (nearest-neighbor) matching method. This table presents the results of OLS regressions of the effect of CEO social capital, on the level of discretionary accruals used in a firm in the reduced samples predicted by propensity score (nearest-neighbor) matching method. The dependent variable (*Dis Accruals*) is defined as the error term estimated by the cross-sectional Modified Jones (1991) Model. CEO social capital is measured by Degree in columns (1) - (3), Eigenvector in columns (4) - (6), and PCA in columns (7) - (9). All regressions include control variables as Real EM (real earnings management estimated by Roychowdhury (2006) Model), Size (natural log of total asset), BTM (Book-to-market equity ratio), ROA (return on asset), Leverage (total debt scaled by total asset), ROAstd (rolling standard deviation of return on assets for the past three years including the current year), CFstd (rolling standard deviation of operating cash flow on assets for past three years including current year), Cycle (days of Operating cycle divided by one thousand), Sales Growth (sales growth in year t), Big4 (indicator variable that equals one if audited by big four auditors), *Duality* (indicator variable that equals one if a CEO also serves as the board chair in year t), and *Tenure* (numbers of years a CEO is in the position). All regressions include time & industry fixed effects, and the errors are robust to firm heteroscedasticity. T-value is reported in parentheses. Statistical significance of the coefficients is designated as ***, **, and * at 1%, 5%, and 10% levels, respectively.

F	2
_	<u>٦</u>
\sim	-

Sample	Entire Reduced Sample	Dis Accruals > 0	Dis Accruals < 0	Entire Sample	Dis Accruals > 0	Dis Accruals < 0	Entire Sample	Dis Accruals > 0	Dis Accruals < 0
Dependent Variable					Dis Accruals				
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
Degree	0.000368**	-0.0000520	0.000739**						
	(2.22)	(-0.41)	(2.44)						
Eigenvector				0.000223*	0.0000143	0.000417*			
				(1.87)	(0.16)	(1.79)			
PCA							0.00592**	0.0000541	0.0106**
							(2.52)	(0.03)	(2.31)
Real EM	0.122***	0.0518***	0.111***	0.119***	0.0505***	0.114***	0.120***	0.0502***	0.116***
	(11.75)	(6.61)	(5.60)	(11.40)	(6.61)	(5.39)	(11.42)	(6.52)	(5.64)
Size	-0.0182***	-0.0101***	-0.0210***	-0.0174***	-0.0109***	-0.0182***	-0.0187***	-0.0104***	-0.0206***
	(-5.76)	(-6.97)	(-3.33)	(-6.08)	(-7.89)	(-2.89)	(-5.95)	(-7.18)	(-3.11)
BTM	-0.00241	-0.00734***	-0.00165	-0.00247	-0.00651***	-0.000619	-0.00238	-0.00704***	-0.00106
	(-1.25)	(-3.24)	(-0.59)	(-1.36)	(-3.06)	(-0.23)	(-1.28)	(-3.17)	(-0.39)
ROA	0.608***	0.215***	0.568***	0.577***	0.215***	0.531***	0.591***	0.213***	0.557***
	(10.14)	(6.07)	(6.76)	(10.01)	(6.27)	(6.16)	(10.19)	(6.27)	(6.66)
Leverage	0.0681***	0.0312***	0.0969***	0.0630***	0.0276***	0.0970***	0.0643***	0.0280***	0.0953***
	(4.59)	(2.89)	(3.88)	(4.41)	(2.64)	(3.86)	(4.43)	(2.68)	(3.79)
ROAstd	-0.0674*	0.0561**	-0.172*	-0.0836**	0.0538*	-0.150**	-0.0793**	0.0579**	-0.154**
	(-1.75)	(2.23)	(-1.95)	(-2.21)	(1.88)	(-2.06)	(-2.09)	(2.15)	(-2.01)
CFstd	0.259***	0.245***	0.117	0.239***	0.268***	0.131	0.250***	0.251***	0.137
	(3.84)	(4.54)	(0.99)	(3.48)	(4.97)	(1.15)	(3.65)	(4.64)	(1.20)
Cycle	-0.00385**	-0.00124	-0.00386	-0.00324	0.000286	0.0248	-0.00135	0.000257	0.0210
	(-2.03)	(-0.97)	(-0.13)	(-1.55)	(0.39)	(0.79)	(-0.71)	(0.22)	(0.70)
Sales Growth	0.00118	0.00565*	-0.00452	0.00193	0.00565**	-0.00574	0.00104	0.00652**	-0.00492
	(0.23)	(1.79)	(-0.49)	(0.48)	(2.47)	(-0.68)	(0.22)	(2.11)	(-0.59)
Big4	0.00246	-0.00611	0.0171**	0.000972	-0.00533	0.0169**	0.00231	-0.00612	0.0164*
	(0.49)	(-1.40)	(2.02)	(0.20)	(-1.24)	(2.01)	(0.47)	(-1.42)	(1.96)
Duality	0.00000797	0.00378	-0.00932	0.000926	0.00455	-0.00579	-0.000690	0.00339	-0.00750
	(0.00)	(1.17)	(-1.61)	(0.27)	(1.41)	(-1.01)	(-0.20)	(1.06)	(-1.30)
Tenure	-0.000477	-0.000734***	-0.000262	-0.000620*	-0.000785***	-0.000185	-0.000420	-0.000677**	0.0000104
	(-1.43)	(-2.70)	(-0.48)	(-1.90)	(-2.93)	(-0.34)	(-1.29)	(-2.54)	(0.02)
_cons	0.0178	0.110***	-0.0343	0.0293	0.112***	-0.0269	0.0497*	0.110***	0.0189
	(0.67)	(4.66)	(-0.77)	(1.14)	(4.96)	(-0.60)	(1.69)	(4.68)	(0.34)
N	10659	7230	3429	10892	7376	3516	10859	7369	3490
adj. R-sq	0.366	0.221	0.506	0.356	0.223	0.481	0.364	0.225	0.497
Industry Fixed Effect included	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Year Fixed Effect Included	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes

t statistics in parentheses * p<.10,**p<.05,***<0.1

Table 8. CEO Social Capital and the use of discretionary accruals: Information vs Reputation Channels (Effects). This table presents the results of OLS regressions of the information and reputation effects of CEO social capital, on the level of discretionary accruals used in a firm. The dependent variable (Dis Accruals) is defined as the error term estimated by the cross-sectional Modified Jones (1991) Model. CEO social capital is measured by Information Channel in columns (1) - (3), and Reputation Channel in columns (4) - (6). All regressions include control variables as *Real EM* (real earnings management estimated by Roychowdhury (2006) Model), Size (natural log of total asset), BTM (Book-tomarket equity ratio), ROA (return on asset), Leverage (total debt scaled by total asset), ROAstd (rolling standard deviation of return on assets for the past three years including the current year), CFstd (rolling standard deviation of operating cash flow on assets for past three years including current year), Cycle (days of Operating cycle divided by one thousand), Sales Growth (sales growth in year t), Big4 (indicator variable that equals one if audited by big four auditors), *Duality* (indicator variable that equals one if a CEO also serves as the board chair in year t), and Tenure (numbers of years a CEO is in the position). All regressions include time & industry fixed effects, and the errors are robust to firm heteroscedasticity. T-value is reported in parentheses. Statistical significance of the coefficients is designated as ***, **, and * at 1%, 5%, and 10% levels, respectively.

Sample	Entire Sample	Dis Accruals>0	Dis Accruals<0	Entire Sample	Dis Accruals>0	Dis Accruals<0			
Dependent Variable			Dis Accruals						
	(1)	(2)	(3)	(4)	(5)	(6)			
Information Channel	0.00444*	0.000229	0.0124***						
	(1.77)	(0.12)	(2.70)						
Reputation Channel				0.00994***	0.00405	0.0104*			
				(2.62)	(1.13)	(1.66)			
REM	0.121***	0.0539***	0.109***	0.122***	0.0542***	0.111***			
	(12.78)	(7.67)	(5.74)	(12.78)	(7.66)	(13.27)			
Size	-0.0180***	-0.0119***	-0.0184***	-0.0178***	-0.0121***	-0.0168***			
	(-7.02)	(-10.08)	(-3.21)	(-7.64)	(-11.03)	(-8.48)			
BTM	-0.00299*	-0.00607***	0.000511	-0.00294*	-0.00601***	0.000382			
	(-1.71)	(-3.10)	(0.19)	(-1.68)	(-3.06)	(0.15)			
ROA	0.562***	0.212***	0.525***	0.562***	0.213***	0.524***			
	(11.15)	(7.90)	(6.65)	(11.21)	(7.89)	(43.60)			
Leverage	0.0735***	0.0368***	0.0965***	0.0751***	0.0373***	0.0990***			
	(5.24)	(3.75)	(4.05)	(5.27)	(3.76)	(7.36)			
ROAstd	-0.0564*	0.0478**	-0.0982	-0.0562*	0.0477**	-0.0980***			
	(-1.74)	(2.17)	(-1.61)	(-1.73)	(2.16)	(-5.11)			
CFstd	0.234***	0.236***	0.0728	0.233***	0.236***	0.0710*			
	(3.80)	(5.18)	(0.71)	(3.78)	(5.18)	(1.85)			
Cycle	-0.00218	0.000815	0.00908	-0.00207	0.000848	0.00754			
	(-0.95)	(0.99)	(0.31)	(-0.91)	(1.02)	(0.48)			
Sales Growth	0.00418	0.00803***	-0.00617	0.00413	0.00802***	-0.00634***			
	(1.06)	(3.03)	(-0.76)	(1.05)	(3.02)	(-3.33)			
Big4	0.00287	-0.00687*	0.0180**	0.00328	-0.00687*	0.0191***			
	(0.64)	(-1.81)	(2.38)	(0.73)	(-1.81)	(2.71)			
Duality	0.00119	0.00370	-0.00646	0.000152	0.00327	-0.00771			
	(0.37)	(1.25)	(-1.20)	(0.05)	(1.08)	(-1.35)			
Tenure	-0.000443	-0.000735***	-0.000124	-0.000631**	-0.000775***	-0.000564			
	(-1.54)	(-3.00)	(-0.26)	(-2.00)	(-3.14)	(-1.05)			
_cons	0.0386	0.118***	-0.00950	0.0381	0.119***	-0.0142			
	(1.48)	(5.40)	(-0.21)	(1.50)	(5.50)	(-0.39)			
N	12847	8762	4085	12847	8762	4085			
adj. R-sq	0.341	0.220	0.460	0.341	0.220	0.459			
Industry Fixed Effect included	Yes	Yes	Yes	Yes	Yes	Yes			
Year Fixed Effect Included	Yes	Yes	Yes	Yes	Yes	Yes			

CEO Social Capital and Discretionary Accruals

t statistics in parentheses

* p<.10,**p<.05,***<0.1

Table 9. Future Profitability, the use of discretionary accruals, and CEO Social Capital. This table presents the results of OLS regressions of CEO social capital and the use of

discretionary accruals on future profitability on a firm. The dependent variable is one-year ahead return on asset (*Future ROA*). The level of discretionary accruals used in a firm (*Dis Accruals*) is defined as the error term estimated by the cross-sectional Modified Jones (1991) Model. CEO Social Capital is measured by *Degree* in column (1) - (3), *Eigenvector* in column (4) - (6), and *PCA* in columns (7) - (9). All regressions include CEO social capital, cash flow from operation (*Operating CF*), the level of non-discretionary accruals (*NA Accruals*), and the level of discretionary accruals (*Dis Accruals*), as well as the interaction terms between CEO social capital and the three variables, to capture the interaction effect of CEO social capital on the variables. Additionally, all regressions include control variables as *Size* (natural log of total asset), *ROAstd* (rolling standard deviation of return on assets for the past three years including the current year), *BTM* (Book-to-market equity ratio, *Sales Growth* (sales growth in year t), and *Big4* (indicator variable that equals to one if audited by big four auditor). All regressions include time & industry fixed effects, and the errors are robust to firm heteroscedasticity. T-value is reported in parentheses. Statistical significance of the coefficients is designated as ***, **, and * at 1%, 5%, and 10% levels, respectively.

CEO Social Capital and Discretionary Accruals

Sample	Entire Reduced Sample	Dis Accruals > 0	Dis Accruals < 0	Entire Reduced Sample	Dis Accruals > 0	Dis Accruals < 0	Entire Reduced Sample	Dis Accruals > 0	Dis Accruals < 0
Dependent Variable					Future ROA				
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
Operating CF	0.825***	0.854***	0.892***	0.841***	0.853***	0.948***	0.839***	0.859***	0.930***
	(18.55)	(15.13)	(12.44)	(15.75)	(14.68)	(9.00)	(16.10)	(14.90)	(9.33)
NA Accruals	0.125*	0.440***	0.129**	0.101	0.442***	0.0722	0.104	0.453***	0.0671
	(1.73)	(9.62)	(2.27)	(1.43)	(9.10)	(0.91)	(1.46)	(9.44)	(0.85)
Dis Accruals	0.157***	0.383***	-0.0230	0.156***	0.395***	0.0175	0.161***	0.398***	0.0203
	(2.67)	(8.43)	(-0.44)	(2.79)	(8.10)	(0.35)	(2.84)	(8.31)	(0.40)
High Degree	0.00793	-0.00248	0.0274						
	(0.66)	(-0.41)	(0.91)						
Operating $CF \times High Degree$	-0.0347	-0.141**	0.0302						
	(-0.39)	(-1.97)	(0.15)						
NA Accruals × High Degree	0.255**	-0.0657	0.140						
	(2.23)	(-0.96)	(1.06)						
Dis Accruals × High Degree	0.262**	-0.0334	0.414***						
0 0	(2.37)	(-0.45)	(2.70)						
High Eigenvector				0.0212**	0.000716	0.0498**			
8 8				(2.37)	(0.12)	(2.14)			
Operating $CF \times High Eigenvector$				-0.0576	-0.123*	-0.0940			
-F				(-0.79)	(-1.67)	(-0.69)			
NA Accruals × High Eigenvector				0.356***	-0.0691	0.387***			
				(3.22)	(-1.04)	(2.90)			
Dis Accruals × High Figenvector				0 294***	-0.0618	0.413**			
				(2.83)	(-0.86)	(2.51)			
High PC A				(2.05)	(0.00)	(2.51)	0.0198**	-0.00283	0.0459**
Ingil I CA							(2.24)	(-0.46)	(2.02)
Operating $CE \times High PCA$							-0.0574	-0.145**	0.0518
Operating CF × High LEA							(0.70)	(2.00)	(0.20)
NA Accruals × High DCA							(-0.75)	(-2.00)	0.418***
NA Acciuais × High FCA							(2.06)	-0.103	(2.02)
Dis Agemula v High DCA							(3.00)	(-1.30)	(5.02)
DIS Accituais × Higii PCA							(2.66)	-0.0708	(2.45)
C'	0.0115***	0.00/01***	0.000***	0.0100***	0.00640***	0.0104***	(2.00)	(-1.07)	(2.45)
Size	0.0115****	0.00081	0.0208	0.0109****	0.00640****	0.0194	0.0112****	0.00672****	0.0194***
DO A-t-1	(8.12)	(5.51)	(0.09)	(8.29)	(5.54)	(6.40)	(8.02)	(5.01)	(0.21)
KOAsid	-0.01/1	-0.00/16	-0.0460	-0.0137	-0.00862	-0.0224	-0.0158	-0.00760	-0.0291
	(-0.65)	(-0.30)	(-0.92)	(-0.53)	(-0.36)	(-0.45)	(-0.61)	(-0.32)	(-0.60)
BIM	-0.00349	-0.00/56**	0.00138	-0.00370	-0.00/3/**	0.000786	-0.00343	-0.00/4/**	0.00149
	(-1.31)	(-2.23)	(0.35)	(-1.41)	(-2.15)	(0.20)	(-1.29)	(-2.21)	(0.37)
Sales Growth	-0.00410	-0.00419*	-0.00654	-0.00438	-0.00418*	-0.00649	-0.00432	-0.00422*	-0.00682
	(-1.26)	(-1.92)	(-0.95)	(-1.38)	(-1.92)	(-1.00)	(-1.36)	(-1.95)	(-1.04)
big4	-0.00407	-0.00542	0.00425	-0.00350	-0.00467	0.00355	-0.00295	-0.00476	0.00628
	(-1.03)	(-1.35)	(0.50)	(-0.90)	(-1.15)	(0.40)	(-0.75)	(-1.18)	(0.71)
_cons	-0.0523***	-0.0184	-0.138***	-0.0554***	-0.0167	-0.147***	-0.0575***	-0.0181	-0.147***
	(-3.80)	(-1.29)	(-4.69)	(-3.64)	(-1.18)	(-4.41)	(-3.72)	(-1.26)	(-4.37)
N	12935	8833	4102	12935	8833	4102	12935	8833	4102
adj. R-sq	0.516	0.538	0.532	0.519	0.537	0.533	0.519	0.539	0.533
Industry Fixed Effect included	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Year Fixed Effect Included	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes

t statistics in parentheses * p<.10,**p<.05,***<0.1

Table 10. Probability of Future Dividend Increase, Accruals-based Earnings Management (Modified Jones (1991) Model), and CEO Social Capital. This table presents the results of logic regressions of CEO social capital and the use of discretionary accruals on the probability of future dividend increase on a firm. The dependent variable is the probability of dividend increase one-year ahead (Future DV Increase). The level of discretionary accruals used in a firm (Dis Accruals) is defined as the error term estimated by the cross-sectional Modified Jones (1991) Model. CEO Social Capital is measured by Degree in column (1) - (3), *Eigenvector* in column (4) - (6), and *PCA* in columns (7) - (9). All regressions include CEO social capital, cash flow from operation (Operating CF), the level of non-discretionary accruals (NA Accruals), and the level of discretionary accruals (Dis Accruals), as well as the interaction terms between CEO social capital and the three variables, to capture the interaction effect of CEO social capital on the variables. Additionally, all regressions include control variables as Size (natural log of total asset), ROAstd (rolling standard deviation of return on assets for the past three years including the current year), BTM (Book-to-market equity ratio, Sales Growth (sales growth in year t), and Big4 (indicator variable that equals to one if audited by big four auditor). All regressions include time & industry fixed effects, and the errors are robust to firm heteroscedasticity. T-value is reported in parentheses. Statistical significance of the coefficients is designated as ***, **, and * at 1%, 5%, and 10% levels, respectively.

CEO Social Capital and Discretionary Accruals

Sample	Entire Reduced Sample	Dis Accruals > 0	Dis Accruals < 0	Entire Reduced Sample	Dis Accruals > 0	Dis Accruals < 0	Entire Reduced Sample	Dis Accruals > 0	Dis Accruals < 0
Dependent Variable				F	uture DV Increase				
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
Operating CF	8.071***	7.988***	9.675***	8.059***	8.532***	8.550***	7.998***	8.201***	8.995***
	(11.72)	(9.36)	(8.38)	(11.86)	(9.89)	(7.02)	(11.50)	(9.51)	(7.33)
NA Accruals	5.830***	5.506***	8.781***	5.354***	4.833***	8.750***	5.471***	4.979***	8.829***
	(8.64)	(6.91)	(5.36)	(7.86)	(6.03)	(5.18)	(7.91)	(6.10)	(5.24)
Dis Accruals	5.668***	4.810***	9.580***	5.543***	4.406***	10.02***	5.563***	4.522***	9.548***
	(8.93)	(6.27)	(5.22)	(8.82)	(5.90)	(5.17)	(8.77)	(6.00)	(5.09)
High Degree	0.179	0.177	0.361*						
	(1.49)	(1.12)	(1.79)						
Operating CF × High Degree	0.892	0.958	1.841						
	(0.75)	(0.63)	(1.15)						
NA Accruals × High Degree	1.076	-0.389	4.175**						
	(0.96)	(-0.31)	(1.97)						
Dis Accruals × High Degree	0.902	-0.848	5.079**						
	(0.85)	(-0.68)	(2.19)						
High Eigenvector				0.0990	0.209	-0.0713			
				(0.82)	(1.32)	(-0.35)			
Operating CF × High Eigenvector				0.969	-0.243	4.564***			
				(0.79)	(-0.16)	(2.73)			
NA Accruals × High Eigenvector				1.934*	0.957	4.860**			
				(1.71)	(0.77)	(2.21)			
Dis Accruals × High Eigenvector				1.110	0.0331	4.670*			
				(1.02)	(0.03)	(1.90)			
High PCA							0.183	0.225	0.305
							(1.51)	(1.41)	(1.51)
Operating CF × High PCA							1.074	0.520	3.247**
							(0.90)	(0.34)	(1.99)
NA Accruals × High PCA							1.751	0.726	4.372**
							(1.56)	(0.58)	(1.99)

$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	Tight C/Y							0.105	0.225	0.505
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $								(1.51)	(1.41)	(1.51)
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	Operating $CF \times High PCA$							1.074	0.520	3.247**
$\begin{array}{c c c c c c c c c c c c c c c c c c c $								(0.90)	(0.34)	(1.99)
$\begin{array}{c c c c c c c c c c c c c c c c c c c $	NA Accruals \times High PCA							1.751	0.726	4.372**
$\begin{array}{c c c c c c c c c c c c c c c c c c c $								(1.56)	(0.58)	(1.99)
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	Dis Accruals × High PCA							1.106	-0.140	5.336**
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$								(1.03)	(-0.11)	(2.22)
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $	Size	0.371***	0.368***	0.372***	0.395***	0.395***	0.385***	0.378***	0.380***	0.359***
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		(18.72)	(15.89)	(9.35)	(20.48)	(17.42)	(10.05)	(19.20)	(16.43)	(9.22)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	ROAstd	-13.79***	-15.30***	-6.855***	-13.83***	-15.38***	-6.617***	-13.83***	-15.36***	-6.624***
$ \begin{array}{c c c c c c c c c c c c c c c c c c c $		(-11.95)	(-12.53)	(-4.05)	(-12.00)	(-12.56)	(-3.62)	(-11.95)	(-12.56)	(-3.81)
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$	BTM	-0.139***	-0.198***	-0.0518	-0.144***	-0.203***	-0.0607	-0.141***	-0.201***	-0.0551
$ \begin{array}{cccccccccccccccccccccccccccccccccccc$		(-3.85)	(-3.11)	(-0.96)	(-3.93)	(-3.15)	(-1.06)	(-3.86)	(-3.14)	(-0.99)
$\begin{array}{c ccccccccccccccccccccccccccccccccccc$	Sales Growth	-0.0902	-0.0209	-0.400	-0.0905	-0.0224	-0.416	-0.0878	-0.0209	-0.386
$\begin{array}{cccccccccccccccccccccccccccccccccccc$		(-0.39)	(-0.29)	(-1.53)	(-0.39)	(-0.30)	(-1.49)	(-0.39)	(-0.29)	(-1.43)
$\begin{array}{cccccccccccccccccccccccccccccccccccc$	big4	-0.211***	-0.267***	-0.121	-0.189**	-0.236***	-0.127	-0.206***	-0.255***	-0.145
$-1.927^{***} -2.142^{***} -1.624^{***} -2.079^{***} -2.388^{***} -1.528^{***} -1.982^{***} -2.268^{***} -1.487^{***}$		(-2.79)	(-2.97)	(-0.81)	(-2.51)	(-2.63)	(-0.86)	(-2.72)	(-2.83)	(-0.97)
	_cons	-1.927***	-2.142***	-1.624***	-2.079***	-2.388***	-1.528***	-1.982***	-2.268***	-1.487***
(-5.87) (-4.79) (-2.99) (-6.22) (-5.22) (-2.73) (-5.98) (-4.98) (-2.73)		(-5.87)	(-4.79)	(-2.99)	(-6.22)	(-5.22)	(-2.73)	(-5.98)	(-4.98)	(-2.73)
N 12935 8827 4102 12935 8827 4102 12935 8827 4102 12935 8827 4102	N	12935	8827	4102	12935	8827	4102	12935	8827	4102
adj. R-sq 0.278 0.269 0.332 0.277 0.268 0.333 0.278 0.269 0.334	adj. R-sq	0.278	0.269	0.332	0.277	0.268	0.333	0.278	0.269	0.334
Industry Fixed Effect included Yes	Industry Fixed Effect included	Yes								
Year Fixed Effect IncludedYesYesYesYesYesYesYesYes	Year Fixed Effect Included	Yes								

t statistics in parentheses * p<.10,**p<.05,***<0.1

Table 11. Stock Return, Accruals-based Earnings Management (Modified Jones (1991) Model), and CEO Social Capital. This table presents the results of OLS regressions of CEO social capital and the use of discretionary accruals on future stock return of a firm. The dependent variable is the cumulative stock return for the twelve-month period ending three months after the fiscal year end (Stock Return). The level of discretionary accruals used in a firm (Dis Accruals) is defined as the error term estimated by the cross-sectional Modified Jones (1991) Model. CEO Social Capital is measured by *Degree* in column (1) - (3), *Eigenvector* in column (4) - (6), and *PCA* in columns (7) - (9). All regressions include CEO social capital, cash flow from operation (Operating CF), the level of non-discretionary accruals (NA Accruals), and the level of discretionary accruals (Dis Accruals), as well as the interaction terms between CEO social capital and the three variables, to capture the interaction effect of CEO social capital on the variables. Additionally, all regressions include control variables as Size (natural log of total asset), ROAstd (rolling standard deviation of return on assets for the past three years including the current year), BTM (Book-to-market equity ratio, Sales Growth (sales growth in year t), and Big4 (indicator variable that equals to one if audited by big four auditor). All regressions include time & industry fixed effects, and the errors are robust to firm heteroscedasticity. T-value is reported in parentheses. Statistical significance of the coefficients is designated as ***, **, and * at 1%, 5%, and 10% levels, respectively.

CEO Social Capital and Discretionary Accruals

61

Sample	Entire Reduced Sample	Dis Accruals > 0	Dis Accruals < 0	Entire Reduced Sample	Dis Accruals > 0	Dis Accruals < 0	Entire Reduced Sample	Dis Accruals > 0	Dis Accruals < 0
Dependent Variable					Stock Return				
	(1)	(2)	(3)	(4)	(5)	(6)	(7)	(8)	(9)
Operating CF	0.376***	0.407***	0.419***	0.401***	0.443***	0.456***	0.404***	0.435***	0.456***
	(5.15)	(3.82)	(4.50)	(5.34)	(3.99)	(4.98)	(5.45)	(3.96)	(5.02)
NA Accruals	-0.0601	0.123	0.103	-0.0399	0.212	0.118	-0.0578	0.153	0.0809
	(-0.86)	(0.76)	(1.12)	(-0.59)	(1.26)	(1.40)	(-0.86)	(0.93)	(0.98)
Dis Accruals	0.0889	0.354**	-0.282**	0.0795	0.441**	-0.259**	0.0992	0.380**	-0.221*
	(1.27)	(2.13)	(-2.28)	(1.16)	(2.55)	(-2.27)	(1.47)	(2.25)	(-1.91)
High Degree	-0.0167	-0.0255	0.00277						
	(-0.88)	(-0.90)	(0.09)						
Operating $CF \times High Degree$	0.119	0.0339	0.195						
	(1.32)	(0.27)	(1.45)						
NA Accruals \times High Degree	0.264**	0.0995	0.250						
	(2.18)	(0.36)	(1.56)						
Dis Accruals \times High Degree	0.158	0.0360	0.368**						
	(1.56)	(0.13)	(2.39)	0.00717	0.00004	0.0205			
High Eigenvector				0.00/1/	0.00804	0.0286			
				(0.38)	(0.29)	(0.97)			
Operating CF × High Eigenvector				0.0547	-0.0489	0.0813			
NA Assessable High Eisensessates				(0.59)	(-0.38)	(0.59)			
NA Accruais × High Eigenvector				0.225*	-0.145	0.205			
Die Alemanie - High Eigenergeten				(1.81)	(-0.54)	(1.58)			
Dis Accruais × High Eigenvector				0.192*	-0.194	(2.56)			
High DC A				(1.62)	(-0.72)	(2.50)	0.000254	0.0150	0.0260
lightCA							-0.000234	-0.0130	(0.86)
Operating CE × High PCA							(-0.01)	(-0.53)	0.0883
Operating CI × High FCA							(0.44)	-0.0420	(0.62)
NA Accruak × High PCA							0.261**	0.00534	0.368**
With Certains × Then I City							(2.04)	(0.02)	(2.14)
Dis Accruals × High PCA							0.145	-0.0388	0.307*
Dis Recitatis A High F Cri							(1.34)	(-0.14)	(1.93)
Size	-0.0148***	-0.0155***	-0.00915	-0.0183***	-0.0206***	-0.0110	-0.0162***	-0.0171***	-0.0115
522	(-3 64)	(-3.14)	(-1.26)	(-4 70)	(-4.29)	(-1 59)	(-4.05)	(-3.49)	(-1.60)
ROAstd	0.101**	0.0945	0.0620	0.105**	0.0926	0.0889	0.103**	0.0940	0.0800
	(2.12)	(1.44)	(0.97)	(2.21)	(1.42)	(1.40)	(2.15)	(1.43)	(1.28)
BTM	-0.0282**	-0.0328*	-0.0202	-0.0281**	-0.0326*	-0.0207	-0.0280**	-0.0326*	-0.0199
	(-2.31)	(-1.86)	(-1.21)	(-2.31)	(-1.86)	(-1.25)	(-2.30)	(-1.86)	(-1.20)
Sales Growth	0.00873	0.00887	0.00322	0.00855	0.00876	0.00297	0.00858	0.00875	0.00331
	(1.39)	(0.99)	(0.47)	(1.36)	(0.98)	(0.44)	(1.36)	(0.98)	(0.50)
big4	0.0306*	0.0136	0.0818***	0.0275*	0.0107	0.0792***	0.0297*	0.0130	0.0807***
	(1.90)	(0.71)	(2.74)	(1.71)	(0.56)	(2.67)	(1.85)	(0.68)	(2.70)
cons	-0.119*	-0.182**	-0.143	-0.104	-0.163*	-0.140	-0.116*	-0.176**	-0.140
_	(-1.75)	(-2.03)	(-1.34)	(-1.53)	(-1.82)	(-1.33)	(-1.70)	(-1.96)	(-1.32)
N	12935	8833	4102	12935	8833	4102	12935	8833	4102
adj. R-sq	0.264	0.259	0.282	0.263	0.259	0.281	0.263	0.259	0.280
Industry Fixed Effect included	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes
Year Fixed Effect Included	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes	Yes

t statistics in parentheses * p<.10,**p<.05,***<0.1